

Received 16 July 2023, accepted 22 August 2023, date of publication 24 August 2023, date of current version 30 August 2023. Digital Object Identifier 10.1109/ACCESS.2023.3308523

## **TOPICAL REVIEW**

# **Power Converters Coolant: Past, Present, Future,** and a Path Toward Active Thermal Control in **Electrified Ship Power Systems**

ALI MOGHASSEMI<sup>01</sup>, (Graduate Student Member, IEEE), S. M. IMRAT RAHMAN<sup>1</sup>, (Graduate Student Member, IEEE), GOKHAN OZKAN<sup>(D1,2</sup>, (Senior Member, IEEE), CHRISTOPHER S. EDRINGTON<sup>001</sup>, (Senior Member, IEEE), ZHEYU ZHANG<sup>1</sup>, (Senior Member, IEEE), AND PHANI KUMAR CHAMARTHI<sup>©1</sup>, (Member, IEEE) <sup>1</sup>Holcombe Department of Electrical and Computer Engineering, Clemson University, Clemson, SC 29634, USA <sup>2</sup>Energy Innovation Center, Clemson University, Clemson, SC 29634, USA

Corresponding author: Ali Moghassemi (amoghas@clemson.edu)

This work was supported by the U.S. Office of Naval Research (ONR) was Authorized for Public Release under Award N00014-22-1-2558 and Award DCN# 543-330-23.

**ABSTRACT** Power converters have widespread applications in automotive, renewables, and power systems. The demand for power modules with low power consumption and high efficiency has increased due to advancements in semiconductor devices. So, power converters need to be highly efficient to reduce costs associated with energy dissipation and cooling requirements. This paper discusses various active thermal control methods for high-power power converters. It covers modulation and configuration techniques, ranging from single configurations to cascaded, modular, and multilevel converters. These concepts form the basis of power electronics building blocks, particularly relevant in all-electric ship systems. Power electronics building blocks represent a thriving technology that will advance ship power systems, the thermal design of which plays a crucial role in managing high heat dissipation levels. Hence, thermal management is essential for reliable device performance. The paper thoroughly studies different active thermal control methods and their impact on power semiconductor devices and converters, categorized per configurations, power routing methods, modulation, and control layers. The review then moves to thermal control methods for the PEBBs concept using multilevel converters in all-electric ship systems. The paper eventually outlines future research directions for the thermal aspect of power electronics building blocks.

**INDEX TERMS** Power semiconductor device, active thermal control, multilevel converter, power electronics building block, modulation strategy, power routing, health monitoring, all-electric ship.

NON	<b>MENCLATURE</b>	
21	NDC	т

3L-NPC	Three-level Neutral-Point-Clamped	CAN	Controller Area Network.
JL-NIC JL VSC	Two level Voltage Source Converter	CHB	Cascaded H-Bridge.
ZL-VSC	All Electric Shin	CHB-MLC	Cascaded H-Bridge Multi-level
AES ALE	All-Electric Ship.		Converter.
ALE	Active Electrine Extension.	СМ	Condition Monitoring.
ALI-00 DI WW	Width Modulation	CONV-60° DPWM	Conventional 60° Discontinuous
ANPC	Active Neutral-Point-Clamped		Pulse Width Modulation.
	Converter	CTE	Coefficients of Thermal Expansion.
		DAB	Dual Active Bridge.
The associate editor	coordinating the review of this manuscript and	DCCS	Distributed Communication and

ATC

The associate editor coordinating the review of this manuscript and approving it for publication was Jiajie Fan<sup> $\square$ </sup>.

Active Thermal Control.

Control System.

DFIG	Doubly-Fed Induction Generator.
DPWM	Discontinuous Pulse Width
	Modulation.
ELT	Estimated Lifetime.
EMI	Electromagnetic Interference.
ESD	Energy Storage Device
FSS	Energy Storage System
EtherCAT	Ethernet for Control Automation
LuciCAI	Technology
EV	Electric Vehicle
	Electric vehicle.
FC-MLC	Flying Capacitor Multi-level
	Converter.
FCS-MPC	Finite Control Set Model Predictive
	Control.
FDDI	Fiber Distributed Data Interface.
FEM	Finite Element Method.
FPGA	Field-Programmable Gate Array.
FRT	Fault Ride-Through.
GaN	Gallium Nitride.
GSC	Grid Side Converter.
HTSSM	High-Temperature Superconducting
	Synchronous Machine.
HVDC	High-Voltage Direct Current.
HVRT	High-Voltage Ride-Through.
IGBT	Insulated-Gate Bipolar Transistor.
IM	Induction Machine.
IMU	Impedance Measurement Unit.
iPEBB	Integrated Power Electronics Building
	Block
LVDC	Low-Voltage Direct Current
LVDC	Low Voltage Dide Through
	Motion And Control Ding Onticol
MEA	Mono Electric Aircraft
	Mole Electric Alician.
MLC	Multi-Level Converter.
MMC	Modular Multi-level Converter.
MOSFET	Metal Oxide Semiconductor Field
	Effect Transistor.
MPPT	Maximum Power Point tracking.
MVAC	Medium-Voltage Alternating Current.
NLM	Nearest Level Modulation.
NPC	Neutral-Point-Clamped.
NPC-MLC	Neutral-Point-Clamped Multi-level
	Converter.
ONR	Office of Naval Research.
OPT-ZSSPWM	Optimal Zero Sequence Injection Pulse
	Width Modulation.
PEBB	Power Electronics Building Block.
PESNet	Power Electronics System Network
PI	Proportional-Integral
PMSM	Permanent Magnet Synchronous
	Motor
PROFIBILS	Process Field Bus
PROFINET	Process Field Net
I NOTINE I DDOEINET IDT	Process Field Not Loophronous Des <sup>1</sup>
FROFINET IKI	Time
DCD	nime.
PSD	Power Semiconductor Device.

PWM	Pulse Width Modulation.
QAB	Quadruple Active Bridge.
RSC	Rotor Side Converter.
RUL	Remaining Useful Lifetime.
SE	Sync Error.
SERCOS	Serial Real-time Communications
	System.
SHDCM	Superconducting Homopolar Direct
	Current Machines.
Si	Silicon.
SiC	Silicon Carbide.
SoH	State-of-Health.
SPI	Serial Peripheral Interface.
SVD	Space Vector Diagram.
SVM	Space Vector Modulation.
SVPWM	Space Vector Pulse Width Modulation.
SyCCo-Bus	Synchronous-Converter-Control-Bus.
ST	Smart Transformer.
THD	Total Harmonic Distortion.
THIPWM	Third Harmonic Injected Pulse Width
	Modulation.
TSEP	Temperature-Sensitive Electrical
	Parameter.
VSC	Voltage Source Converter.
WBG	Wide Bandgap.
WRN	White Rabbit Network.
ZCS	Zero-Current Switching.
ZVS	Zero-Voltage Switching.

Photovoltaic.

ΡV

## I. INTRODUCTION

#### A. BACKGROUND

Power converters find application in a diverse range of fields, including but not limited to renewable energy systems like photovoltaic systems [1], electric vehicles (EVs) and motor drives [2], [3], custom power devices [4], all-electric ship (AES) power system [5], and so on. Reliability and efficiency of converters have been demanding issues, as they play a crucial role in the design of power systems [6], [7], [8]. Power semiconductor devices (PSDs) are typically integrated into modules to increase power density and reduce material consumption [9]. The failure distribution among power electronic components is depicted in Figure 1a. The figure indicates that capacitors and semiconductors stand out as the most fragile components in terms of vulnerability among power electronic components. Figure 1b compares the underlying reasons for stresses and failures in power electronics converters. As can be seen, temperature is the most dominant cause of stress in power electronics converters, highlighting the importance of controlling the temperature of PSDs [10]. This also highlights the crucial role of PSDs in ensuring the reliability of power systems. PSD failure is the underlying cause of power electronics converter faults, indicating that enhancing the reliability of PSDs results in improved power converter reliability.



#### **B. MOTIVATION**

As mentioned, one can declare that the power system reliability is strongly linked to the reliability of the PSDs. PSD failure is the fundamental reason for power converter faults, meaning that improving the reliability of the PSDs leads to an increase in the power converter reliability. To address this issue, several studies have been conducted over the years, including condition monitoring (CM), active thermal control (ATC), and remaining useful lifetime (RUL) estimation methods, as shown in Figure 2. The rationale behind the CM is to utilize physical measurements to detect the occurrence of failures or degradation in the entire power system. Based on the outputs of CM, appropriate actions can be taken to prevent a sudden shutdown of the system.

Another approach to enhance the reliability of the PSDs is the ATC. As shown in Figure 2, the degradation indicator data obtained from online CM methods can be utilized to passively update and, significantly, actively control the system's lifetime through the use of ATC. As mentioned above, thermal stress is the primary cause of PSD failures. ATC mitigates the thermal stress on the components by reducing the amplitude of temperature fluctuations or lowering the average temperature without modifying the converter. This means there probably be no additional expenses for improving the converter design or elements. It is worth pointing out that the thermal control capability and the power system's performance must be simultaneously considered. Finally, the RUL estimation method is, by and large, deployed to design the ATC and then validate its effect. Figure 2 shows the relationship between the method mentioned earlier; CM, ATC, and RUL estimation.



FIGURE 2. Relationship between the CM, the ATC, and the RUL estimation methods.

The material layers in the power converter modules serve to provide both functionality and electrical insulation [12]. In general, aluminum bond wires are used for the electric connection between chips and terminals [13]. Figure 3 depicts the layer sequence of an insulated gate bipolar transistor (IGBT) module tied to a heatsink. Due to ambient temperature and power consumption variations, the layers undergo continuous heating and cooling processes, referred to as thermal cycles [14]. Due to the mismatch in coefficients of thermal expansion (CTE) between the layers, mechanical stress occurs in the module [15], leading to aging and damage to the modules [16]. The magnitude of the thermal cycles contributes to the degree of aging [17]. Short-term cycles mainly cause the bonds' fatigue, leaving bond wire on the surface and rising. Long-term cycles, despite the high thermal capacitance of the base plate, affect its temperature and lead to solder fatigue [18]. Plus, the modules' failure means inevitable maintenance, replacement, and thus outages and the operators' expenses.

Various studies have been conducted to improve the lifetime and reliability of modules, including developing improved connection technologies and assembly methods. In [9], sintering is proposed as an alternative to the chips soldering to improve the strength of the connections. The matching of CTE values and the heat dissipation performance is enhanced with the help of advanced materials [9], [14]. The delamination of the substrate can be reduced by avoiding 90° angles in the pattern, as discussed in [14]. Regulated cooling can mitigate low-term ambient temperature swings, as proposed in [19]. Several methods are deployed for cooling the electronic elements [20], [21]. These approaches are categorized into two main groups, active and passive, as per the fact that an energy resource is needed for the cooling process. Their merits and demerits are provided in [22], [23], and [24]. In the ATC approach, a fluid is normally applied as the coolant, and based on the working fluid's type, a fan or a pump is used to produce the cooling flow and forced convection. Turning to the passive techniques, which are traditional cooling methods, heat sinks, and fins are employed where free convection is the dominant mechanism of heat transfer [25].

ATC is focused on reducing the impact of short and medium-term thermal cycles. The approach uses temperature-related control parameters to adjust the junction



FIGURE 3. Schematic of an IGBT module [32].

temperatures in real time. Its objective is to minimize thermal stress on the module by reducing temperature fluctuations. Thermal control aims to regulate the junction temperatures by adjusting the losses in the targeted chips through specific temperature-related control parameters. These parameters include the switching frequency [26], [27], modulation technique [28], DC-link voltage [29], gate voltage [30], or distributing reactive power [31]. For instance, a short-term temperature drop can be averted or decreased when losses are escalated temporarily by increasing the switching frequency. While CM necessitates understanding the junction temperature, ATC can function without it. Additionally, an electrothermal model can derive real-time estimations of junction temperatures through electrical measurements, enabling more accurate thermal stress management [26].

## C. STATE OF CONTRIBUTIONS

Numerous scholars have conducted research on the ATC of power electronics converters, as documented in the literature. A few papers have reviewed the ATC for PSDs and power converters through cooling strategies. Although prior works have proposed various techniques to improve the ATC, CM, and RUL aspects of power electronics converters, there are no solutions to address the ATC of power electronics building blocks (PEBBs) concept, health monitoring of PEBBs, and their aging/degradation. This is particularly important in AESs, where the modular multilevel converter (MMC) based PEBBs concept needs cost-effective ATC methods to reduce maintenance costs and increase reliability/availability. To maintain power converters, it is necessary to anticipate the state-of-health (SoH) and potential damage progression to implement appropriate thermal control measures. The main focus of this research is to provide a detailed overview of various ATC methods applied in different configurations of power electronics converters, including single converter, cascaded or modular converter, and parallel converter configurations. Additionally, the impact of ATC on power module and converter performance is analyzed, with a particular emphasis on reliability, power loss, and junction temperature reduction. Finally, because modular and multilevel converters shape PEBB stacks, the thermal controls of which need to be designed, the paper introduces the concept of PEBBs, specifically focusing on their application in AESs. Furthermore, the scope of future work for ATC of PEBBs in various high-power applications is proposed.

This work will serve as a comprehensive and useful reference for health monitoring, degradation, aging, and thermal management of the PEBBs concept with the help of previous works on power converter topologies in terms of their configuration, modulation strategy, and the most important parameters that are useful for thermal control aspect of PEBBs in AESs.

#### **D. STRUCTURE OF PAPER**

The remaining sections of the paper are structured in the following manner. Section II presents various ATC methods categorized by converter configurations (single, cascaded or modular, and parallel) and various ATC methods of power routing for modular converters. Section III provides the ATC methods and various control levels with different bandwidths, such as the ATC methods at the modulation level (including switching loss reduction and conduction loss redistribution), the system function level (including active power or energy storage and reactive power), and the other levels. Section IV provides the discussion, challenges, and scope of future work in the AES power system. This section starts with the introduction of PEBBs concept, the thermal aspect of the PEBBs concept and moves to the discussion on ATC for various topologies related to the PEBBs concept (such as multilevel converters, multi-phase converters, and soft-switched converters). Then it presents ATC methods in the PEBBs concept, the communication protocols of PEBBs, and how they can help ATC methods. After that, this section details the advantages and disadvantages of different converters used for shaping high-power rating converters in the PEBBs concept. Ultimately, this section finishes with thermal data acquisition, health monitoring and degradation, cost analysis, ATC's effect on the system's operation, and PSD technology and packaging aspects for future work. Finally, Section V concludes the paper.

#### **II. ATC METHOD**

Before reviewing the ATC methods, a brief explanation of the thermal model used in the literature review should be provided.

#### A. THERMAL MODEL

Improvement of thermal management systems can be achieved with a highly accurate estimation of semiconductor device temperature. To do so, high-fidelity thermal models must be developed. There are several approaches to developing thermal models, which are discussed below. The measured response from real systems can be adapted to fit the parameters of thermal equivalent circuits. Two standard forms of the thermal circuit have been extensively used for thermal modeling, as shown in Figure 4: Cauer and Foster networks [33]. The following section provides an overview of these two thermal networks.





## 1) CAUER NETWORK

The heat conduction path of a semiconductor material can be modeled by an equivalent electrical circuit, as presented in Figure 4a. This circuit is called the Cauer network [34]. Thermal resistance over the heat conduction path is denoted by R, and C denotes thermal capacitance. The junction and ambient temperatures are indicated by  $T_i$  and  $T_a$ , respectively. The power dissipation is denoted by P. The Cauer network represents the physical properties of the heat conduction path more accurately, which is a significant advantage. Moreover, this network allows independent modeling of the heat sink and power semiconductor device, as shown in Figure 4b. Later they can be combined to develop intricate thermal models [36]. Nevertheless, the Cauer network is not precisely accurate due to heat spreading. Also, providing exact parameters to the model is a complex process. The advantage of Foster networks is that they are easy to use [35]. However, the use of the Foster network is restricted to specific applications only. And unlike the Cauer network, the elements of the Foster network cannot be designed individually and combined later.

## 2) FOSTER NETWORK

Foster networks are a non-parametric model tuned to step response data of the thermal conduction path. There is no physical significance of the components in this circuit, and each component only represents a time constant. A typical foster network is presented in Figure 4c. The advantage of Foster networks is that they are easy to use [35]. However, the use of the Foster-type circuit is restricted to specific applications only. And unlike the Cauer model, the elements of the Foster network cannot be designed individually and combined later.



**FIGURE 5.**  $\Delta T_i$  control in PWM method [46].

It should be noted that authors in the literature review used various thermal networks with different accuracy and layers depending on the focus of their work (see Figure 3). The thermal modeling of power converters and related mathematical equations used in each paper deserve a separate review paper which is out of this work's scope. Since this paper only focuses on the ATC methods, this work does not review the thermal networks in the literature. Table 1 compares the various thermal models for complexity, accuracy, and multichip interfaces [40]. More details and mathematical models of thermal networks can be found in [36] and [40].

## **B. ATC IN DIFFERENT CONVERTER CONFIGURATIONS**

The ATC, as a newly-presented method to modify power losses and thermal stress, will be represented in this section. The central concept behind this problem is to adjust the power converter's control variables related to temperature to modify the junction temperature and reduce damage caused by thermal cycling, as discussed in [27] and [32]. The ATC enhances the reliability of PSDs, thereby increasing the power system's lifetime. The primary objective has been to regulate junction temperature and other related-temperature values. Based on the type of converter, one can categorize the ATC into single, cascaded, and parallel converters. Single converter configurations include 2L and 3L converters in AES and drive applications, as well as converters in solar applications, as discussed in [41] and [42]. Cascaded converter configurations consist of cascaded h-bridge (CHB) converters and modular multi-level converters (MMC) [43], [44]. Lastly, parallel 2L and 3L DC/DC converter configurations have been used in wind energy and motor drive applications.

#### 1) ATC IN SINGLE CONVERTER CONFIGURATIONS

One clear-cut way to adjust the switching frequency is to realize thermal control, which directly affects the power losses [27], [45], and a negligible effect on the power system's working condition. In [46], a method was presented to reduce the switching frequency of a 2L converter in an adjustable speed drive based on changes in junction temperature. As shown in Figure 5, there are three stages to control  $\Delta T_j$ . First, average power losses are calculated based on power factor and commanded switching frequency  $(f_s^*)$ . Second, the temperature variation  $(\Delta T_i^*)$  of the PSD

TABLE 1. Comparison of the most commonly used thermal networks.

Thermal Network	Complexity	Individual Layer	Accuracy	Experimental Input	Multi-chip Interface	Notes
Foster [35]	Simplest	No	Less accurate at high fre- quency	No	No	Commonly used in in- dustry
Cauer [34]	Less complex	Yes	Accurate	No	No	Commonly used
Finite element method (FEM) [37]	Less complex	N/A	N/A	No	Modeled	high computation bur- den
Modified Elmore [37]	Complex	Yes	More accurate	Specific points	Modeled	Combines FEM and two RC parallel cir- cuits, Less computa- tion
Modified FEM [38]	Complex	Yes	Most accurate	Specific points	Modeled	Combines FEM and four RC parallel cir- cuits, Less computa- tion
Fourier [39]	Complex	Yes	N/A	No	Modeled	Thermal diffusion, high computation burden

Note. N/A: Information is Not Available in the paper.

is calculated. Third, the switching frequency is determined using a hysteretic control [46]:

$$fs = \begin{cases} f_s^{min} & \Delta T_j^* > T_1, \\ f_s^* & \Delta T_j^* < T_2, \\ unchange & T_1 < \Delta T_j^* < T_2 \end{cases}$$
(1)

In the above equation,  $T_1$  and  $T_2$  represent the upper and lower boundaries for the hysteretic temperature variations  $(\Delta T_j^*)$ , and  $f_s^*$  is commanded switching frequency. They adjust the switching frequency, which is calculated based on the average temperature  $(T_{jm})$  and the temperature variation  $(\Delta T_j^*)$ . While the results indicate that this control method can enhance reliability more effectively than controlling a single parameter, combining control parameters can increase the computational complexity of the control system.

Modifying the modulation strategies and applying modern control approaches is an excellent way to address the above issue. In [47], a switching control method based on pulse width modulation (PWM) is proposed for managing power losses in the three-level neutral-point-clamped (3L-NPC) converter. The technique described here is active lifetime extension (ALE) and does not require additional equipment. It can be applied to the modulation index (m) range between 0.5 and 1. In this inverter, there exist 27 various switching states in the 3L-NPC, as displayed in Figure 6. Applying different switching states can reduce switching losses or conduction losses. For instance, the region highlighted in light yellow in Figure 6 remarks redundant switching states. Should the conduction losses need to be reduced, the switching states resulting in higher conduction losses are eradicated from the switching sequence. Otherwise, if the reduction of switching losses has priority, the corresponding states are prohibited, as depicted in Figure 7.

The use of different switching states can help to decrease switching or conduction losses. The light yellow section in Figure 6 represents redundant switching states. If it is



FIGURE 6. Various kinds of switching states in the 3L-NPC inverter [47].



**FIGURE 7.** The effect of redundant states on switching and conduction losses [47].

necessary to decrease conduction losses, the switching states that result in higher conduction losses are eliminated from the switching sequence. On the other hand, if reducing switching losses must be decreased, the related states are prohibited, as depicted in Figure 7.

The inner hexagon region highlighted in light turquoise in Figure 6 contains redundant switching states that can be utilized to alter the current paths flowing in the PSDs. This alteration helps lower the conduction or switching losses depending on the priority. This control strategy is particularly useful for grid-connected power electronics converters during fault ride-through (FRT) scenarios, such as high-voltage ridethrough (HVRT) and low-voltage ride-through (LVRT), and during start-up operation of drives, where the modulation index is not high, and the voltage reference is in the inner hexagon of Figure 6, as presented in [48], [49], and [50].

The study presented in [51] proposes a junction temperature controller for a 2L-3P converter based on the finite control set model predictive control (FCS-MPC). The controller aims to control the amplitude of thermal cycles and estimates the load current, junction temperature, and leading to thermal stress for all space vectors of the next sampling instant. The cost function parameters are then obtained per these predictions, including the error from the current reference, the thermal stress across the switches, the temperature discrepancy between the chips, and the PSDs' total power losses. After that, the parameters are weighed, and the space vector concerning the lowest cost function is selected and applied to the converter. Figure 8 provides a visual representation of this approach. According to this method, a control sequence for a current-source active rectifier was presented in [52] as a control algorithm based on a finite number of switching states of the power converter. To opt for the optimal switching state, an objective function calculates the error between foreseen and reference values of both electrical and thermal objectives. The results show that the electrical and thermal objectives can be attained by minimizing the multiobjective weighted cost function. It is worth pointing out that the power converter's output performance might decrease, and the computation's complexity might increase.



FIGURE 8. Schematic of ATC method based on FCS-MPC [52].

In the study conducted in [52], the authors proposed a control algorithm for a rectifier based on a finite set of switching states of the power converter. The algorithm aims to opt for the best switching state by calculating an objective function (which is based on electrical and thermal objectives) that measures the discrepancy between predicted and reference figures. The findings indicate that the multi-objective weighted cost function can be optimized for the purposes.

It bears pointing out that the power converter's output performance may decrease, and the computation's complexity may rise due to the algorithm's implementation.

## 2) ATC IN CASCADED OR MODULAR CONVERTER CONFIGURATIONS

Modular or cascaded power converters are becoming more prevalent in different voltage ranges. As previously mentioned, the most commonly used topologies for modular power converters are CHB converters and MMCs. For having many cells and submodules, these converters are susceptible to uneven thermal stress, which can negatively impact the PSDs and the overall reliability of the power converter. Ref. [53] presented a thorough review of current state-ofthe-art opportunities and the future perspective of MMCs for transportation electrification applications.

In [54], a power routing method was proposed to implement ATC by unevenly distributing the load among the cascaded or modular configuration modules. As shown in Figure 9, power routing involves optimizing the allocation of power among each module to improve the efficiency and reliability of the system [54]. The modules may be connected in series, parallel, or a series-parallel configuration. When connected in a series, the modules share the same current, but each module has the ability to regulate its output voltage. This allows for customization of the power output of each individual module by adjusting its output voltage. Similarly, when connected in parallel, the modules share the same voltage, but each module can independently control its output current, which provides flexibility in adjusting the current output of each module. To regulate the cell's power, the current is the parameter manipulated instead of the voltage. Therefore, this methodology involves using the power routing technique for the CHB converters [55], a modular smart transformer (ST) that comprises a CHB to convert medium voltage AC to medium voltage DC [56], and dual active bridges (DAB) for converting medium voltage DC to low voltage DC [57]. It should be mentioned that the most damaged cells can be conserved with the help of this method. The power routing will be discussed in detail in the next section.



FIGURE 9. The basis of the power routing method [54].

Turning to MMCs, a wide variety of research works have been done on MMCs to enhance the output performance [58], [59], [60], [61], decrease the complexity of computation [62], [63], [64], and also provide a cost-effective power loss balance among submodules [57], [65], [66]. Hence, the ATC technique is commonly utilized to achieve a uniform



FIGURE 10. The control schematic of the submodule's temperature [70].

distribution of thermal stress across the various submodules of the system, which helps to enhance the lifespan of the power system [57]. For instance, the unbalanced thermal distribution among submodules brought about by the mismatch in the parameters of the submodule was proposed in [67]. Because the capacitors in the submodules are not identical, the switching and conduction losses of the associated submodules will differ, resulting in uneven thermal stress distribution across the submodules. To overcome this issue, an active thermal balancing control method was proposed, which involves integrating the lower IGBT's junction temperature and the capacitor's voltage into an algorithm that employs a weight function. The weight factor is varied between zero and a predetermined value to ensure that there is both equal thermal circulation among the submodules and balanced capacitor voltage levels [68]:

$$L_i = (1 - \alpha) \times v_{norm}^i - (\alpha) \times T_{norm}^i \times sing(i_{arm}) \quad (2)$$

In which  $\alpha$  represents the weight factor, and  $v_{norm}^i$  and  $T_{norm}^{i}$  indicate the variance in the capacitor's voltage and the junction temperature, respectively. The experimental study was done in various case studies considering different submodule capacitors. The findings were used to validate the suggested thermal balancing control techniques that enable uniform distribution of temperature across submodules [67]. Reference [69] presents an approach to incorporate the junction temperature into the capacitor voltage balancing algorithm to achieve uniform thermal distribution among the submodules. Unlike the research in [67], the temperature of the devices within the submodule is combined individually with the dedicated capacitor voltage, resulting in different distinct cost functions for the upper/lower IGBTs and upper/lower diodes. The appropriate cost function is chosen for each sampling instant, considering the current direction in each leg and the need to either insert or bypass submodules. The outcomes indicated that the suggested approach significantly reduced the inconsistency and temperature dispersion among the submodules.

In [70], submodule thermal balancing was achieved by adjustment of the capacitor's voltage of each submodule in a leg and simultaneously maintaining the sum of the voltages of the submodule's capacitor at the rated value to control the voltage of the DC-link. Figure 10 illustrates the comparison between the temperature of each submodule, denoted as  $T_{SM,i}$ , and the average temperature of all submodules in the leg denoted as  $T_{avg}$ , where the difference is fed into a proportional-integral (PI) controller. The PI controller generates the voltage differential for each submodule voltage



FIGURE 11. 1P hybrid-based MMC topology [71].

reference, as described in [70]. The capacitors' voltages can be adjusted per the work done in [72], but additional terms related to the temperature were considered. It should be noted that while the temperature among submodules was balanced, the capacitors' unbalanced voltages resulted in a distorted voltage waveform of the multi-level leg.

Figure 11 shows that the thermal stress circulation among the submodules of hybrid-based MMC can become more unbalanced under a high voltage modulation index, as noted in [71]. The ATC was introduced for half- and full-bridge submodules to address this issue. In the case of full-bridge submodules, two kinds of bypassed switching modes were modified to produce a symmetrical switching arrangement when the leg voltage is positive, as depicted in Figure 12 [71]. The same procedure is applied when the voltage of the leg is negative. While the power loss distribution in full-bridge submodules is more balanced, the symmetrical switching arrangement does not negatively affect the performance, decreasing thermal stress on the most highly stressed devices. In addition, a thyristor that can resist high current is connected in parallel with the lower leg of the submodule, as shown in Figure 13, to bypass the positive current of the leg to decrease the thermal stress across the lower IGBT.

#### 3) ATC IN PARALLEL CONVERTER CONFIGURATIONS

A feasible solution is to use a parallel connection of many power converters to address the challenges of applications where the current is high and the output voltage is low.



**FIGURE 12.** The configuration of switching states when the leg voltage is positive [71].



FIGURE 13. Half-bridge submodule with a parallel thyristor [71].



FIGURE 14. The thermal loading control method [73].

This configuration enhances the system's reliability as redundancy can be straightforwardly applied. However, even though sharing the load is used to dispatch the load current and achieve an equal load sharing, it does not ensure an equal thermal stress circulation among the parallel converters. To address this, an active thermal sharing for the parallel DC/DC converter was presented in [73], [74], and [75]. This approach uses the power converters' temperature values to redistribute the load current between parallel converters. Figure 14 shows that the current and temperature data were combined, and the resulting data was utilized to improve load sharing. The purpose of this control diagram is to ensure that the thermal stress is spread evenly. The merit of this method is its ease of implementation. However, it should be noted that there may be a small rise in the failure rate of individual converters.

Implementing a droop-based ATC method, as described in [81] and [82], can enhance the system's reliability. In addition, [76] proposed a control approach for sharing the load in which the droop gain is adjusted per the computed lifetime and calculated as the sum of the used lifetime. This is expressed in the following equation:

$$R_{d,k} = ACL_{k(pu)} \times R_{do} \tag{3}$$

where  $R_{do}$  is the highest acceptable value of the droop gain. To achieve load sharing among the converters based on the thermal stress across the PSDs and ensure equal used lifetime for all converters, it is necessary to regulate the corresponding droop gains in accordance with the used lifetime of each converter. By doing so, the accumulated used lifetime of the converters can be balanced, resulting in the improved overall reliability of the system.

The power routing approach can also be utilized in a parallel configuration to achieve balanced aging of the converter cells. This method was proposed in [77] and [78] for a parallel DC/DC converter and a two-level voltage source converter (2L-VSC) in a permanent magnet synchronous motor (PMSM), respectively. Specifically, it reallocates power to each converter cell based on its aging status, accomplished by regulating the duty cycle to produce the appropriate switching pulses. By extending the lifetime of the most aged cell, the reliability of the entire system is enhanced.

In wind power applications, as depicted in Figure 15, the parallel converter configuration has a drawback of significant temperature variation for the fluctuation of the wind speed. To mitigate temperature fluctuations in the PSDs resulting from changes in wind speed, such as wind gusts, the ATC method was implemented in the wind power system, taking into account reactive power, as shown in Figure 15 [31], [79], [80], and [83]. The delivered reactive power can considerably affect components' loading in this configuration. It also does not affect the converter's existing mechanical/electrical power.

The reactive power can regulate the phase angle (between the output current and voltage). It also alters the current flowing through the PSDs, which will impact the power loss and thermal stress across the PSDs. Using an appropriate amount of reactive power to heat the device when the power is low will significantly reduce the temperature fluctuation of the PSDs. However, it bears noting that using reactive power has some demerits, including being applicable only in parallel converter configurations and increasing the diodes' thermal load.

Table 2 summarizes the most critical ATC methods for power converters based on various configurations.



FIGURE 15. Wind power system with parallel converter configuration [79].

## C. ATC AND POWER ROUTING

Power routing is an advanced active thermal management technique that balances the thermal stress within modular

Config.	Converter	Method	Advantages/Highlights	Disadvantages/Comments
Single Converters	1P H-bridge in- verter [45]	Switching fre- quency control	Reduces thermal swing resulting from power cycling.	Changing switching frequency may in- crease harmonics.
	2L converter [46]	Switching fre- quency control	Adapts switching frequency to junc- tion temperature changes for reduced power losses.	Added 3rd-order harmonic to increase the voltage transfer ratio, but no har- monic result was provided.
	3L-NPC converter [47]	Switching con- trol based on PWM	Reduces switching or conduction losses via space voltage vector redun- dancies.	Increased ripple and potential higher computational burden.
	3L-NPC converter [48] - [49]	SVM	Optimal loss and thermal redistribu- tion under LVRT	The proposed method's dependency on heat sink design and PSD usage can alter thermal distribution.
	2L-3P converter [51]	FCS-MPC	Controls thermal cycles and optimizes power losses through cost function- based selection of space vector.	Computation burden should be ana- lyzed.
	2L-3P converter [52]	FCS-MPC	Optimizes switching states by min- imizing a multi-objective weighted cost function: DC-link control and ATC	Computation burden should be ana- lyzed.
Cascaded or Modular Converters	Modular STs with CHBs [54] - [56]	Power routing	Unevenly distributes load among modules to improve efficiency and re- liability	Efficiency in modular architecture re- quires simultaneous attention.
	Modular STs with CHB & DABs [57]	Power routing	Enhances block lifetime by internally routing power in a modular power converter, including a CHB connected to DABs.	Applicability limited to ST systems not mentioned.
	MMC [67] - [71]	Sorting algorithm	Achieves thermal stress balance by integrating junction temperature and capacitor voltage into control algo- rithms.	Computation burden should be ana- lyzed.
Parallel Converters	DC/DC converter [73] - [76]	Load sharing control	Utilizes temperature values to redis- tribute load current among paral- lel converters, ensuring equal thermal stress distribution.	Possibility of slight increase in indi- vidual converter failure rate.
	DC/DC converter [77] - [78]	Power routing	Achieves balanced aging of converter cells by reallocating power based on aging status.	No studied effects of power routing on harmonics.
	Parallel converter [31], [79] - [80]	Reactive power control	Alleviates temperature fluctuations by utilizing reactive power to regulate phase angle and minimize thermal stress.	Reactive power circulates exclusively within parallel converters to heat up PSDs without being injected into the power grid.

TABLE 2. Summar	y of important ATC	C methods based	d on power conve	rter configurations.
-----------------	--------------------	-----------------	------------------	----------------------

power electronic converters [84]. Through the power routing approach, each modular power electronic converter module manages a different amount of power to mitigate the thermal stress generated by thermal cycling. The primary goal of this method is to prolong the lifetime of the most vulnerable modules in a modular power electronic converter. This is achieved by operating the modules under uneven loading conditions and reducing the stress on the most deteriorated modules by transferring their load to less vulnerable modules.

Figure 16a shows a modular converter with three modules where modules 1 to 3 handle power  $P_1$  to  $P_3$ , respectively. So, the total power will be  $P_T = P_1 + P_2 + P_3$  where  $P_1 = P_2 = P_3$  under regular operating conditions. However, under balanced load-sharing conditions, the aging of the modules will be unequal, which might lead to an untimely breakdown of the modules and, thus, unscheduled maintenance. Figure 16b shows that, with the implementation of power routing, each module handles a different power value depending on its aging status. In this way, power routing can be applied to optimize the maintenance schedule and

VOLUME 11, 2023

increase the lifetime of modular power converters. The cells within a modular converter can be linked in a series, parallel, or hybrid arrangement, as shown in Figures 17a, 17b, and 17c, respectively [85].

In the case of modules connected in series, the output power of each cell can be managed by adjusting the output voltage of each cell. Likewise, for parallel connected modules, the output power from each cell can be regulated by adjusting the output current from each cell. Lastly, for a combined series and parallel structure, the current, voltage, or both can be controlled to regulate the power output from each cell. The primary goal of power routing is to improve the converter's lifetime by optimizing its maintenance schedule. Calculating the converters' RUL and SoH is necessary to apply power routing methods effectively. Then maintenance can be scheduled simultaneously for all the modules of a power electronic converter by considering the RUL of the modules. This concept is illustrated in Figure 18 in which RUL equalization is achieved between two PEBBs. Here the red bars indicate the expected failure time of the two PEBBs,





(b) balancing the power between modules





FIGURE 17. Power routing in different modular converter topologies [54].

and  $P_1$  and  $P_2$  indicate the power each PEBB processes. By applying the power routing method, the power managed by each PEBB can be manipulated so that maintenance for both PEBBs is scheduled at the exact same time. The green bar represents the unified maintenance schedule for both PEBBs.

The power routing method can be employed for modular converters to regulate the temperature of the PSDs and increase their RULs. Several approaches have been investigated in literature for implementing power routing techniques with series and parallel connected converters, as discussed in the sections below.

#### 1) POWER ROUTING IN PARALLEL-CONNECTED MODULAR CONVERTER

A closed-loop controller has been implemented in [77] that considers the SoH of a three-cell parallel-connected DC/DC converter to improve the maintenance cost of the system,



FIGURE 18. RUL management via power routing [84].



FIGURE 19. Power routing in three-cell DC/DC boost converter [77].

as shown in Figure 19a. Figure 19b shows the control scheme for such a converter and the simulated results.

Here, the SoH block estimates the aging of each cell by utilizing data on the usage of the power system and instantaneous electrical measurements. Then it provides a temperature reference  $(T_n^*)$  to the thermal controller block, determining the appropriate power-sharing factor  $(k_n^*)$  for the parallel cells. Afterward, the reference power for each individual cell  $(P_n^*)$  is determined by dividing the total instantaneous reference power  $(P^*)$  by the control factor  $(k_n^*)$ . From the reference power, it is possible to determine the reference current  $(I_n^*)$ , which can be utilized to calculate the voltage  $(v_n)$ that must be modulated by each individual cell. A conventional PI controller assists in determining  $v_n$ . The proposed



FIGURE 20. Lifetime estimation of three-cell DC/DC boost converter [86].

algorithm successfully redistributes the power among the cells and achieves thermal management.

A power routing mechanism is presented in [86] that decreases the aging imbalance in an interleaved boost converter with three modules. Figure 20a depicts the proposed control algorithm. The upper segment of the control scheme determines the power that needs to be processed by each module to re-balance the RUL of the modules. The bottom segment of the control scheme employs two PI controllers to determine the reference power  $P_n^*$  and reference voltage for the power electronic devices. The application of power routing allows proper thermal management of the cells and extends the converter's overall lifetime, as shown in Figure 20b. By the utilization of the suggested power module has been extended by 24 months.

A virtual resistance-based power routing technique for parallel converters of the low voltage stage of an ST is presented in [56]. The power routing method is utilized when the power system is under partial load. The arrangement of the 3-stage ST for power routing is depicted in Figure 21a. The LV side consists of three parallel connected converters. The proposed control framework for implementing the power routing method is depicted in Figure 21b. Here, three virtual resistors ( $R_{v1}$ ,  $R_{v2}$ , and  $R_{v3}$ ) are used to distribute power among the three converters. The values of resistors are adjusted based on the feedback obtained from the CM and are determined as a function of the cumulative damage:

$$R_{vi} = f(D_{acc}) \tag{4}$$

The accumulated damage is estimated by sensing the semiconductor device's junction temperature  $(T_j)$ . A path with high resistance indicates low loading, and vice versa for a



**IEEE**Access



(a) three-stage ST archit



(b) virtual power routing method



(c) the ELT of parallel converters with and without power routing



(d) the effect of power routing on the junction temperature of the converters

FIGURE 21. A virtual power routing technique for parallel converters in an ST [56].

low-resistance path. The reference current for any specific converter  $(i_{cn}^*)$  can be calculated by using the total reference current  $(i_{ref}^*)$  generated by the voltage controller and the

virtual resistors as expressed below [56]:

$$[i_{c1}^{*}, i_{c2}^{*}, i_{c3}^{*}] = \left(\frac{i_{ref}^{*}}{R_{\nu 2}R_{\nu 3} + R_{\nu 1}R_{\nu 2} + R_{\nu 1}R_{\nu 3}}\right)$$
$$.[R_{\nu 2}R_{\nu 3}, R_{\nu 1}R_{\nu 2}, R_{\nu 1}R_{\nu 3}]$$
(5)

Figure 21c depicts the converters' estimated lifetime (ELT) with different initial damages. The converters will break down after different intervals if power routing is not applied. The ELT of the most aged cell in the system can be extended with the virtual resistor-based power routing technique. Figure 21d illustrates the effect of the suggested power routing approach on the converters' junction temperature. Initially, the load on converter number 3 is less than the other two converters, and after t = 20s, converter number 3 is loaded more than the other converters.

The graph shown in Figure 21d illustrates that the virtual resistors cause the current to vary, and as a result, the junction temperature changes with the loading of the converter. When a converter is not in use, the junction temperature decreases, but when the load on the converter increases, the junction temperature rises again. A power routing algorithm for a more electric aircraft (MEA) with four parallel DABs has been proposed in [87] and [88] to improve the system's lifetime. The MEA architecture and the control algorithm are shown in Figures 22a and 22b, respectively. The power routing algorithm presented in this study estimates the RULs of the converters by analyzing the junction temperature profile generated by the converters' mission profile. The total damage for each converter can be calculated as follows [88]:

$$D_i = D_{ini,i} + \sum_{period=1}^m \Delta D_i \tag{6}$$

where  $D_i$  is total accumulated damage and  $D_{ini,i}$  is initial damage. The RUL for a specific converter can be expressed below [88]:

$$LT_{exp,i} = \frac{1 - D_i}{\Delta D_i} \tag{7}$$

where  $\Delta D_i$  is the rate of damage change over a specific time interval. Due to thermal cycling, the RUL for each converter is different, and the controller determines the power each converter must process to extend the lifetime of the converter. During the full load operation, all converters process an equal amount of power, while the power routing algorithm is solely operational under partial load conditions. Figure 23a illustrates the accumulated damage of the converters, both with and without the implementation of power routing. The cells wear out at a different rate when no lifetime control method is activated. The cells process power as per their thermal stress with active lifetime control. Thus, the accumulated damages converge to a common point, as seen in Figure 23b.

The effect of power routing on the aging of the ST's ten parallel connected DAB cells is investigated in [89]. The finding from this study is depicted in Figure 24. Figure 24a illustrates the damage sustained by the DAB cells while sharing the power equally for 400 months. The first breakdown



FIGURE 22. Power routing for an MEA [88].

occurs in the 200th month, and the last occurs in the 370th month, leading to several interruptions in regular operation due to maintenance intervals. However, while the power routing algorithm is implemented, the breakdown of all cells occurs in the 293rd month (Figure 24b), which reduces the maintenance intervals.

A case study has been presented in [54] to demonstrate the effectiveness of power routing in reducing the thermal cycling of the most deteriorated cells of the low-voltage conversion stage of an ST. Figure 25 shows the results from this study, where cell 1 was the most deteriorated cell. The proposed power routing technique reduces the thermal cycling in cell one, which results in the extension of the lifetime of cell one. The efficacy of the virtual resistors-based power routing method in reducing thermal stress in a modular DC/DC converter, in which several quadruple active bridges (QABs) converters are included, has been investigated in [90]. The outcomes obtained indicate the efficiency of the control algorithm in postponing the system failure.

## 2) POWER ROUTING IN SERIES-CONNECTED MODULAR CONVERTER

Modulation strategies that have been modified can be utilized to apply power routing in modular converters that are connected in series and parallel. These strategies enable the



FIGURE 23. Impact of power routing on the accumulated damage [88].



Power in Cells Junction Temperature in Cells 100 90 90 80  $p_1 = p_2 = p_3$  $T = T_i$ 85 No Power 80 Routing 70 60 75 (%) 70 65 õ 50 40 POWOR 60 55 30 20 50 10 0 45 40 Time Time 100 90 90 80 70 60 85 80 . i.cell2 p 75 (%) Κŧ  $p_3$ 70 ô 50 40 65 Power 60 30 55 Power Routing 20 10 50 45 (Unload Cell 1) 40 Time Time

FIGURE 25. Thermal cycle reduction in the most aged cell of the LV stage converter in an ST [54].



FIGURE 26. Power routing with DPWM method [91].

FIGURE 24. The impact of power routing on the aging of DAB cells [89].

control of the thermal stress of each converter individually. A discontinuous PWM (DPWM) based power routing technique for CHB converters is presented in [91]. The DPWM

reduces the switching loss by clamping the output voltage at either the positive or negative DC-link voltage. The loss reduction is significant compared to the continuous PWM



FIGURE 27. Modular ST [92].



FIGURE 28. Junction temperature of CHB and DAB cells with and without clamping signal [92].

techniques because the semiconductor devices cannot switch in the clamping region. As shown in Figure 26a, the proposed method has two modulation signal types. A positive offset voltage  $V_{off,p}$  is added to the fundamental reference voltage  $V_{ref,fund}$  to obtain the modulation signal  $V_{ref,ns}$ , which has a clamping zone. The most vulnerable converters are modulated with  $V_{ref,ns}$  as the clamping region reduces the generated losses. The other modulation signal  $V_{ref,s}$  has no clamping region and a negative offset voltage  $V_{off,n}$ , compensating the clamping of other converters. The less vulnerable converters are modulated with this signal. The ultimate goal of the modulation strategy is to delay the failure of the most vulnerable converters. The modulation strategy's application is depicted in Figure 26b, depending on the converters' condition. The converter with lesser RUL is modulated with  $V_{ref,ns}$  to decrease the thermal stress, and the one with higher RUL is modulated with  $V_{ref,s}$ , compensating for the positive offset voltage.

The power routing concept presented in [91] is extended in [92] that applies a discontinuous modulation technique to the medium-voltage alternating current (MVAC) to lowvoltage direct current (LVDC) building block of an ST to achieve thermal management (Figure 27). This study considers the DAB as the DC/DC converter. After applying the proposed modulation technique, the estimated CHBs' and DABs' junction temperatures with a fixed 60° clamping angle are shown in Figures28a and 28b, respectively. For the CHB cell, the junction temperature rises to 82°C with the non-clamped signal, while for the clamped one, the junction temperature reaches around 43.5°C. For the DAB cells, the maximum junction temperature reaches 99.8°C with the clamped signal and 78.7°C for the non-clamped signal.

An advanced discontinuous modulation technique for implementing power routing in STs has been proposed in [93] that manages the thermal stress within the CHB converters and the isolated DC/DC converters. Another power routing technique for improving the thermal performance of CHB converters has been proposed in [94]. This work has applied a non-conventional phase-shifted PWM that increases the most vulnerable cells' lifetime and reduces the cells' harmonic distortion. A multi-frequency modulation-based power routing technique is presented in [55], where a third harmonic component is injected into the duty cycle of CHB converters. The injection of a third harmonic component maximizes the utilization of DC-link voltage, akin to the third harmonic injected PWM (THIPWM) method. Figure 29 illustrates a 3-cell CHB converter featuring three paths  $(P_A, P_B, \text{ and } P_C)$ for power flow. Failure of any single cell or power path will stop the operation of the converter. One approach to avoid this situation is to decrease the load on the power path with the highest degradation level. Figure 29b depicts a scenario where the loading on power path A is reduced to protect cell A from untimely breakdown.

Figure 29 demonstrates the concept of unbalanced powersharing between a three-cell CHB where the first three rows show the duty cycle of the cells, and the fourth row shows the power-sharing between the cells. In Figure 30a, the powersharing is balanced between the cells, with each cell having the same duty cycle. Figure 30b shows unbalanced power sharing between the cells where the cells with higher loads have a higher duty cycle. Here, the unbalanced power sharing is restricted when the duty cycle becomes maximum. The power imbalance capacity is extended when the third harmonic voltage is injected with the fundamental voltage, as shown in Figure 30c.

## **III. ATC METHOD AND CONTROL LEVELS**

The failure procedures of power converter elements are not straightforward and rely on several decisive factors [10], [68], [95], [96], [97]. Thermal cycling, which refers to temperature variations of the power converter, is a common cause of



(b) 3-cell CHB with unbalanced power-sharing

FIGURE 29. Power flow between CHB cells in case of balanced and unbalanced power-sharing [55].



FIGURE 30. Power routing technique [55].

failures in converters. Temperature fluctuations on various materials with different CTE can bring about continuous interruptions in the contacting areas within or between the elements, ultimately leading to a shortened lifespan of the entire power converter. A multitude of producers of power converter elements, like the PSD or capacitors, have modified their lifespan approaches, which are more than capable of making an accurate and reliable estimation of the lifespan as per specific thermal behaviors seen by the elements [12], [98], [99], [100]. A research study conducted in the 1990s aimed to establish a numerical correlation between

the thermal cycling features of the IGBT, the quantity of failed thermal cycles, the rise in the temperature fluctuation  $dT_i$ , and the average temperature value  $T_m$ . The study found that the lifetime of PSDs decreases as the temperature fluctuation and mean temperature level increase, as shown in Figures 31a and 31b, respectively. This correlation was described and validated based on several analytical models and lifespan tests in [12], [101], [102], [103], and [104]. What is more, other determining factors; such as the extensive usage of power electronics, network, and electric machine faults, short-circuits and open-circuits, inrush current, etc.; can have a negative role in thermal loading and hence bring about overheating damages to the converter. In general, the power determines the voltage or current of the converter. The variation in the loading of power converter components can hugely influence the variable mission profiles, especially in applications like motor drives, photovoltaic systems, or wind turbine systems, where the power is inconsistent. This leads to complex thermal behaviors and burning-out failures of the elements [105].





The thermal cycling in each device varies depending on the place and reasons for the failure. As a case in point, an empirical temperature measurement study on an opened IGBT using an infrared camera is depicted in Figure 32, where the IGBT's real-time temperature in a 10kW 3P-3L-NPC photovoltaic (PV) inverter is presented. The configuration of the power converter is illustrated in Figure 33 [106]. The chip temperature  $T_j$  and base plate temperature  $T_c$  are two testing points for measurement. The thermal profiles of these measured temperatures are represented in Figure 32a, in which the converted power alters as the solar irradiance changes. Figure 32a illustrates that the temperature of the junction ( $T_j$ ) is higher than that of the case ( $T_c$ ) when there are bigger magnitudes



FIGURE 32. Testing outcomes of thermal behaviors inside the PSD [106].

of thermal variations and slow variations in temperature due to power changes. The thermal behavior of  $T_j$  and  $T_c$  over a shorter time period of 0.2s is shown in Figure 32a, where the solar irradiance maintains consistent, and the converter is loaded at a nominal power of 10kW. Another type of thermal cycling is shown in Figure 32b, where  $T_j$  fluctuates faster at a frequency of 50Hz with a smaller but steady magnitude.



FIGURE 33. Configuration and testing conditions for figure 32 [106].

It can be concluded that the lifespan and reliability of a converter rely heavily on the thermal loading of the PSDs. Various aspects of the energy conversion system's operations can affect the converter's thermal feature. Consequently, the thermal stress across the PSDs can vary significantly as per their places and the disturbances for the converter's load, resulting in different behaviors and time constants. By and large, the failure of an individual element can occur due to a discrepancy between the stress level, which refers to the thermal loading or cycling of the device, and the strength level, which is represented by the device's intrinsic ability or rating to endure thermal stress. According to Figure 34, the thermal stress experienced by a PSD relies heavily on the user's behavior and mission profile, making it preferable to have a range of stress levels instead of a single point of focus. Furthermore, the strength of individual elements must have a distribution degree to account for variations in production and installation into the system. Consequently, an intersection area between the stress and strength levels can be recognized, which indicates a likelihood of failure.



FIGURE 34. ATC for enhanced reliability of power converters.

One practical approach to reduce the probability of failure or increase the reliability of power electronic elements is to minimize the crossing area between the stress and strength levels, which can be achieved by reducing the stress range. As shown in Figure 34, this can be accomplished by decreasing the thermal loading of the PSD by minimizing fluctuations or reducing the mean temperature. Thus, the converter's strength does not require to be altered, and no extra hardware cost is required to improve the strength of elements.

As mentioned earlier, several aspects with different time constants influence a converter's thermal behavior. Therefore, there are ample opportunities and degrees of freedom to modify the thermal loading of the PSDs, resulting in more desirable reliability characteristics. Due to the relatively high power loss density and intense thermal cycling, PSDs are considered one of the most fragile elements in power electronics converters [10], [68], [97].

The control layers and variables in a converter provide various possibilities to change the thermal behavior of the PSDs. Figure 35 depicts the different control levels, with the gate driver being the lowest. At this level, the voltage and current of the drive and gate resistance can be adjusted to revise the PSD's switching power loss and temperature. The bandwidth at this level is relatively high, around a few microseconds. The modulation level can be used to control thermal behavior by adjusting the switching frequency, as shown in Figure 35. This method can modify the switching and conduction losses of the PSDs without affecting the converter's performance. The bandwidth at this level depends on the switching frequency, around ten to hundreds of microseconds. At higher levels, thermal control can be achieved by changing the output voltage, the output current, the DC-link voltage, the active power, and the reactive power. The bandwidth at this level is quite low, around a few milliseconds.



FIGURE 35. Various control layers to manage the thermal behavior of the power converters [107].

## A. ATC AT MODULATION LEVEL

Switching control methods in power converters influence the total harmonic distortion (THD) of the output voltage and current. It also determines the path of the current in the PSDs. Many switching control methods have been advanced to reach higher DC-link voltage, lower loss distribution, and lower THD values [108], [109]. Since the switching control methods can change the PSDs' loading, they can enhance the loss distribution among PSDs, leading to superb thermal loading and reliability. The switching control method's effect on the PSDs' thermal loading can be described according to the switching and conduction loss.

#### 1) MODULATION WITH SWITCHING LOSS REDUCTION

Utilizing improved PWM techniques such as DPWM or declining the carrier frequency can remarkably reduce switching losses [110], [111], [112], [113], [114], [115], [116]. However, a trade-off must be considered, as a reduced number of switching counts can lead to higher current ripples [117]. Declining switching loss in a 2L-VSC is a highly recognized approach, as seen in Figure 36. DPWM is a method that is used to decrease switching loss. It involves clamping the voltage reference of the output to the upper or lower section of the carrier during specific intervals to maintain the related PSDs in their current state without turning ON or OFF, reducing switching loss during that subinterval. The process is depicted in Figure 37. DPWM can decline the total loss of the IGBT, resulting in a lower junction temperature swing. Figure 37b shows that the conduction loss of the device can also be altered with DPWM.



FIGURE 36. The 3P-2L-VSC converter: configuration and control approach [32].



FIGURE 37. The 2L-VSC converter: power loss and junction temperature measurements [32].

The DPWM is not restricted to the use of 2L-VSCs. Multi-level converters can also be applied for this end [118]. Reference [116] presents a case study on a 3L-NPC converter, in which some various DPWM methods based on optimal zero sequence injection PWM (OPT-ZSSPWM), such as conventional 60° DPWM (CONV-60° DPWM) [27], and alternative 60° DPWM (ALT-60° DPWM) [119], are discussed. It was concluded in [116] that using CONV-60° DPWM and ALT-60° DPWM can significantly reduce the thermal stress on the overheated PSD in the converter [116].

## 2) MODULATION WITH CONDUCTION LOSS REDISTRIBUTION

Another approach to modifying the thermal loading is conduction loss, particularly when it comes to switching redundancy like the 3L-NPC converter, as shown in Figure 33 [116]. In a 3-phase converter system with a typical star connection, introducing a common-mode offset to the voltage references does not impact the current. However, it will alter the distribution of conduction loss among the PSDs. The space vector diagram (SVD) of the 3L-NPC converter, as depicted in Figure 38, indicates that all vectors located within the inner hex possess switching redundancies. This provides greater control flexibility to adjust the current paths flowing through the PSDs for better LVRT capability.

A wide variety of modified and optimized thermal modulation methods are represented in [48], [49], and [120]. Figure 39 displays the space vector modulation (SVM) method over one switching period. In this figure, the state vector 1-1-1 is removed, meaning that the converter state's time is declined when current flows via the clamped diode and associated switch. Figure 40 depicts the outcomes of the PSD's temperature in the converter under an extreme LVRT process. The results indicate that an optimized or modified modulation sequence leads to more symmetrical thermal loading and decreased thermal stress on the most stressed PSDs during LVRT.



FIGURE 38. Switching redundancies in the 3L-NPC converter [32].

#### **B. ATC AT SYSTEM FUNCTION LEVEL**

To maintain the desired energy conversion level, this thermal management level is solely suitable for a limited number of power converters, as shown in Figure 41 [32]. Thermal controlling can be attained by recirculating parallel power







FIGURE 40. PSDs' temperatures in the 3L-NPC converter under LVRT [48].



FIGURE 41. Active/reactive power routing to balance thermal stress among converters [32].

converters' active and reactive power. Regardless of some drawbacks, such as higher losses in some overloaded power converters, it is not confirmed whether controlling the PSDs' losses and temperatures in one of the parallel converters is feasible.

## 1) ATC BY REACTIVE POWER

The amount of reactive power a converter can provide is not necessarily limited to the power at the converter's input. It can significantly affect the loading of elements, so it is an amp approach to reach the ATC for enhanced reliability. Figure 42, presented in [31], illustrates the impact of varying reactive power on the loss distribution of the same 3L-NPC converter. The figure shows that reactive power modifies the phase angle between the voltage and current of the output and alters the PSDs' current amplitudes. Furthermore, the reactive power delivered to the power network is closely connected to its power stability and voltage level, particularly when it comes to weak grids with restricted power capacity [121], [122], [123], [124], [125]. Consequently, many grid codes have been established to restrict the amount of reactive power injected by power converters. The mentioned codes can markedly restrict the thermal control capability that can be achieved by utilizing the converter's reactive power.

Figure 43 depicts a doubly-fed induction generator (DFIG) wind turbine system. The DFIG structure allows for the support of reactive power from either the stator side or the grid side converter (GSC) [126], [127]. So, the reactive power that is being distributed among the DFIG system is controllable. Figure 43 shows that the amount of reactive power injected into the power grid remains constant, thereby significantly extending the applicable reactive power for controlling device temperature, regardless of grid codes. However, certain restrictions, such as over-modulation, the current capacity of the PSDs, and DFIG, must also be considered when implementing this control approach [126]. If there are increases in the wind speed, these sudden changes in adverse thermal cycling in the converter. Hence, an appropriate reactive power control scheme in a DFIG system can regulate the swings of junction temperature under such circumstances. Figure 44 illustrates the basic control diagram for this approach [126]. The IEC 61400-1:2019 defines a standard wind gust with a one-year return period. The thermal cycling of power converters with/without thermal reactive power control is shown in Figures 45a and 45b, respectively. Based on Figure 45a, the thermal stress is least severe at a synchronous wind speed due to the lack of active power in the GSC and most severe in the rotor side converter (RSC) due to the small fundamental frequency of the rotor current. The maximum junction temperature fluctuation in the GSC declines from 11°C to 7°C with extra reactive power, as displayed in Figure 45b. However, this fluctuation remains the same at 18°C in the RSC.

Likewise, controlling the PSD's temperature by reactive power is attainable in a parallel converter-based wind system with several converters (see Figure 46). Here, the converter is required to control thrice more power than the wind system depicted in Figure 43, but with the exact generator capacity. To cope with this, parallel converters are generally used, which allow for temperature control of the PSDs by distributing reactive power within the converters [31]. The results are shown in Figure 47 where it can be observed that the temperature swings of the overheated PSD can be significantly decreased, from  $32^{\circ}$ C to  $12^{\circ}$ C.



FIGURE 42. Distribution of losses in the 3L-NPC converter based on various reactive power values [31].



FIGURE 43. Junction temperature regulation in the power converter of a DFIG wind turbine system by implementing reactive power control [126].



FIGURE 44. Thermal control diagram of the converters in the DFIG system [126].

#### 2) ATC BY ENERGY STORAGE OR ACTIVE POWER

The power converter's current magnitude is significantly influenced by the active power, affecting the power semiconductor devices' loss and thermal loading. Unlike reactive power, controlling the available active power in the converter is generally not feasible. However, an energy storage system (ESS) can be used as an additional source of active power [128], thereby enabling this control function. The ESS can be located in the DC-link, as shown in Figure 48, to reduce thermal excursions in the grid-side converter and instabilities in power injected into the grid. However, some challenges associated with using ESS include determining the type of energy storage technology, sizing the ESS appropriately, and evaluating the thermal enhancement provided by the ESS. The ESS comprises an energy storage device (ESD)



FIGURE 45. Thermal cycling of the back-to-back power converters in the DFIG system during abrupt wind speed changes [126].

and a bidirectional power converter. Figure 48 illustrates the power control method for the ESS [128]. Initially, the wind turbine's power production is calculated. Then a high-pass filter is used to obtain short-term power variations that serve as a reference for the ESS's charging power. This helps smooth out the power flow in the GSC and offers independence in controlling the wind power converter from the ESS.

Figures 49a and 49b show the power and temperature characteristics of the power converter when combined with the ESS, respectively. Case 1 has P = 0.05 MW, E = 270 kWh, and the weight is 1.7 t Li-ion. Case 2 has P = 0.177 MW, E = 6.25 kWh, and the weights are 1.25 t Ultracapacitor and 0.12 t Li-ion. Case 3 has P = 0.200 MW, E = 2.6 kWh, and the weight is 0.52 t Li-ion. As per this figure, the power is smoothed with the help of the ESS. Thus, the thermal fluctuation is decreased. What is more, the temperature gets smoother as the ESS becomes larger [128].







FIGURE 47. PSDs' temperatures of the 3L-NPC converter with paralleled converters during abrupt wind speed changes in the DFIG system [126].



**FIGURE 48.** Junction temperature control with the integration of ESS and back-to-back converters in the DFIG system [128].

## C. ATC AT OTHER LEVELS

Over the years, other techniques have also been developed for the ATC of power converters. Active gate control is one such technique that can regulate the losses of PSDs without



FIGURE 49. The GSC as per various energy storage capabilities [128].

affecting their functionality. Adjusting the gate drive voltage allows both losses (conduction and switching) of the PSDs to decline, thereby reducing the thermal cycling magnitude [30]. For example, an active gate driver has been proposed in the literature to compensate for variations in the on-state resistance of IGBTs [129]. A different active gate control has been used to balance currents among parallel-connected IGBTs [130]. Modifying the control parameters of active gate control can also reduce the thermal cycling in the PSDs.

Changing the switching frequency is another effective solution for managing the thermal behavior of power devices. As seen in previous studies, this can be achieved through linear or hysteresis control [26], [27], [28]. The ATCs used in these studies have lowered the switching frequency compared to the uncontrolled system, albeit with an increase in the current ripple [27], [28].

One approach to managing thermal behavior is to vary the modulation index using the DC-link voltage, which affects the loss distribution between IGBTs and diodes [32]. Another method is presented in [27], where a dynamic current limitation is set to prevent overheating and potential damage to the PSD. This allows for low-power operation of nearly overheated systems. In MMCs, specific submodules are subjected to higher thermal stress than others. An algorithm is suggested in [131] to balance MMC submodules' junction temperatures.

It bears noting that almost all ATC methods discussed in the literature are open-loop and do not rely on temperature feedback from the device. This is primarily because temperature measurement within PSDs is not straightforward in real-world applications such as AESs, where high power densities are involved. The thermal model-based estimation is a potential solution, but it heavily depends on accurate models for losses and the thermal behavior of the converter. These models are still not advanced enough to protect the converter's intricate functioning conditions, power device

Table 3 summarizes the most important ATC methods based on various control levels.

they remain areas of active research.

#### IV. DISCUSSION, CHALLENGES, AND SCOPE OF FUTURE WORK IN AES POWER SYSTEMS

packaging technologies, and cooling methods. Therefore,

Several studies have explored open-loop thermal loading controls that do not rely on the device's temperature feedback. This is because measuring a PSD's temperature during the process is challenging, particularly in high-power-density applications such as AES power systems. One potential solution is to use thermal models to predict PSD's temperature, but this method heavily relies on the converter's loss and thermal models' precision. These areas are currently the subject of extensive research as existing models are not yet advanced enough to fully account for the complex operating conditions of the converter, packaging technologies, and cooling requirements and procedures.

Since this work focuses on ATC methods of power electronics converters specifically for AES power systems, this section will first discuss and compare various high-voltage power converters (e.g., multilevel and modular converters) which shape the concept of power electronics building blocks (PEBB) in AES power system applications. It is worth pointing out that although the thermal management issue may initially appear simple, considering the moderate heat generation and available surface area for heat dissipation on each PEBB, there are two challenging constraints to overcome. Firstly, direct liquid cooling is impossible for PEBBs, meaning liquid connections cannot be established. Secondly, the PEBBs must be easily replaceable by the ship's crew while the ship is underway, requiring them to be compact, lightweight, and resistant to physical damage. A large finned air-to-air heat exchanger is out of the question. These constraints pose a significant challenge in transferring heat from the PEBBs to the corridor, as these limitations render many existing electronics cooling technologies ineffective.

Then this section will explain how the ATC methods reviewed so far can be applied and improved in the PEBBs concept for AES power systems. Next, future research opportunities in the ATC methods will be presented because they are pivotal to considering other possible scenarios for future design.

#### A. THERMAL CONTROL OF PEBBS CONCEPT

Newly emerging materials, components, and system concepts; such as wide bandgap (WBG) materials, silicon carbide (SiC) based PSDs, PEBBs, and integrated power systems; have been, are, and will continue to enable future marine systems as different from today's systems as steamships were to sailing ships [132]. Shipboard electric power systems benefit from the progress made in power converters, which are supported by many different elements, subsystems, and system technologies [133]. One notable technology is the PEBB, which was initially conceptualized and proposed

Control Level	ATC Method	$\operatorname{Advantages/Highlights}$	Disadvantages/Comments
Modulation Level	Switching loss re- duction [110] - [116]	Enhanced PWM techniques minimize switching losses but may increase current ripples.	Each method has drawbacks and can- not simultaneously improve all require- ments, e.g., lower switching counts result in higher current ripple.
	Conduction loss re- distribution [48] - [49], [120]	Conduction loss redistribution in the 3L- NPC converter addresses switching redundancy through voltage reference adjustments.	Conduction loss redistribution may in- crease harmonics.
System Func- tion Level	Reactive power [31], [121] - [126]	Reactive power control affects loading and loss distribution. In DFIG wind turbine sys- tems, regulating reactive power injection con- trols junction temperature swings and thermal stress in PSDs.	Reactive power-based controls for junc- tion temperature stabilization are im- pacted by wind gust conditions, wind configurations, and grid code require- ments.
	ESS or active power [128]	Active power affects losses and thermal stress. ESS, as an extra power source in the DC-link, reduces thermal excursions and instabilities.	High-power applications, like AES power systems, have high ESS costs. Integrating ESS and active power control may reduce ESS costs.
Other Levels	Active gate control [30], [129] - [130]	By adjusting gate drive voltage, losses can be regulated without impacting functionality, com- pensating for variations in on-state resistance and balancing currents among parallel IGBTs.	Thermal-related measurement is chal- lenging, especially in AES power systems with high power density.
	Varying switching frequency [27] - [28]	Modifying switching frequency with linear or hysteresis control effectively manages thermal behavior.	Changing switching frequency may in- crease current ripple.
	Varying modulation index [32]	Modulating DC-link voltage impacts loss distribution between IGBTs and diodes.	Not applicable universally, only feasible with a converter-fed dc-link, which limits its applicability.
	Submodule junction temperature balanc- ing [131]	The proposed algorithm equalizes junction tem- peratures within MMCs by addressing elevated thermal stress in specific submodules.	Harmonic study can also be considered.

TABLE 3.	Summary of	important	ATC methods	based on	control levels.
----------	------------	-----------	-------------	----------	-----------------

by the US Office of Naval Research (ONR) in 1998 as a means of enhancing the availability, performance, and cost-effectiveness of converters in future AESs [134], [135]. Implementing the PEBB concept has dramatically impacted the innovation, manufacturing, commissioning, and operation of converters in AES applications [136].

Similar to Figure 35 in which different control layers for managing the thermal behavior of single converters were highlighted, one must understand the control layers of the PEBBs concept to control the thermal behavior of high-voltage power converters in large-scale applications such as AES power systems. These layers are shown in Figure 50 [136]. Figure 51 depicts an AES power system composed of several power electronics converters [137], which has become bigger and more complicated. In this regard, various advancements, including grid configuration and topology, voltage stability analysis, power quality, fault detection, and protection, have been made [138], [139]. To advance thermal management, power ratings, costeffectiveness, availability, and total performance, various cutting-edge topologies over traditional VSCs have been presented. Some of them have already been implemented in the AESs.

#### 1) MULTI-LEVEL CONVERTERS (MLC)

Unlike the basic VSC, MLCs are power converters that generate AC voltages using more than two DC voltage levels. MLCs offer various benefits, including reduced AC ripple, lower voltage stress on both the PSDs and loads, and a higher



FIGURE 50. Control hierarchy in PEBBs concept [136].

effective switching frequency. A great variety of MLCs, such as the NPC multi-level converters (NPC-MLC), the flying capacitor multi-level converters (FC-MLC), the CHB multilevel converters (CHB-MLC), and MMCs have been recently presented [140], [141], [142]. The NPC-MLC is advantageous in terms of high-power density since it has minimal



FIGURE 51. An AES power system with zonal distribution (podded propulsors) [137].

needs for passive components. On the other hand, CHBs and MMCs have modular structures, offer FRT capability, utilize low-voltage PSDs, and possess high efficiency and low quantization on the voltage and current. Still, they require many capacitors and have lower power densities. A thorough study of MLCs and their implementation in various fields, including renewable energy systems, EVs, and charging stations, has been conducted in a comprehensive review published in [143]. The work showed that MLCs offer lower THD values, lower distortion in input current, and lower switching frequency, thereby lower switching losses. On the flip side, having many components results in lower reliability, higher cost, and more complicated controllers.

The authors in [71] presented an ATC method of hybrid MMC under overmodulation operation. The ATC was introduced for half-/full-bridge submodules. Two kinds of bypassed switching modes were modified in full-bridge submodules to produce a symmetrical switching arrangement when the leg voltage was positive. The same procedure is applied when the voltage of the leg is negative. While the power loss distribution in full-bridge submodules is more balanced, the symmetrical switching arrangement does not negatively affect the performance, decreasing thermal stress on the most highly stressed devices. Reference [144] proposed an implementation of the NPC-MLC based on a switchingcell array method, adding redundant conduction paths on one side and more options to distribute the switching losses on the other. This way, the converters' temperature distribution is balanced, reducing thermal stress and improving reliability.

Reference [69] presents an approach to incorporate the junction temperature into the capacitor voltage balancing algorithm, called nearest level modulation (NLM), to achieve uniform thermal distribution among the submodules. Unlike the research in [67], the temperature of the devices within the submodule is combined individually with the dedicated capacitor voltage, resulting in different distinct cost functions for the upper/lower IGBTs and upper/lower diodes. The appropriate cost function is chosen for each sampling instant, considering the current direction in each leg and the need to either insert or bypass submodules. The outcomes indicated that the suggested approach significantly reduced the inconsistency and temperature dispersion among the submodules.

A quasi-two-level FC-MLC for PEBBs was presented in [145]. the results showed that the temperature distribution and the measured heat sink temperature are both improved.

#### 2) MULTI-PHASE CONVERTERS

AES power systems typically utilize power converters that have a 3-phase input or output topology. However, using multi-phase motors with more than three phases can improve propulsion motors' power ratings and FRT capability. In addition to providing modularity and redundancy in the structure, using multi-phase converters can also result in lower common mode voltages, which is advantageous for motors and reduces the need for filters. Reference [147] discussed the modeling and control of VSCs associated with different multi-phase power electronics converter topologies, primarily deployed in multi-phase electrical machines. For instance, multi-phase converters play a crucial role in marine applications, where FRT capability, efficiency, and noise are the pressing issues [148]. Reference [149] studied the progress of modern AESs in terms of propulsion drive motors, such as PMSMs, induction machines (IMs), high-temperature superconducting synchronous machines (HTSSMs), and superconducting homopolar DC machines (SHDCMs).

#### 3) SOFT-SWITCHED CONVERTERS

DC/DC power converters usually have resonant features that enable zero-voltage switching (ZVS) or zero-current switching (ZCS) to decrease switching losses and increase the switching frequency. Some soft-switching topologies for 3P AC power converters have been proposed by incorporating auxiliary circuits like 2L-VSCs, 3L-NPCs, and active neutralpoint-clamped (ANPC) converters [150]. However, despite the advantages of each topology, soft-switched converters have not been widely adopted in practical applications due to the added complexity, cost, and restricted benefits for drives, where high switching frequencies and fast dynamics are not necessary. A soft switching converter for AESs has been proposed and experimentally verified in [151] and [152], respectively. The proposed converter offers high power density and efficiency, minimizing switching power losses.

#### TABLE 4. Advantages and disadvantages of different VSC configurations [140], [146].

Configuration	Advantages	Disadvantages
NPC-MLC	<ul> <li>Modularity in structure</li> <li>Simplicity in structure</li> <li>Good dynamic response</li> <li>Less required DC sources</li> <li>Good FRT capability</li> <li>Lack of floating capacitors</li> <li>Balanced switches' blocking voltage</li> </ul>	<ul> <li>Having restricted DC-link voltage balance to the 3L configuration</li> <li>The higher the voltage level, the higher the clamping diodes are required</li> <li>Unbalanced distribution of losses</li> <li>Inexpensive</li> <li>The higher the clamping diodes, the lower the reliability</li> </ul>
FC-MLC	<ul> <li>Modularity in structure</li> <li>Good FRT capability</li> <li>Low required DC sources</li> <li>Lack of clamping diodes</li> <li>Low voltage stress across the switches</li> <li>The higher the voltage level, the more efficient it is</li> </ul>	<ul> <li>Inexpensive due to the capacitors, as opposed to the diodes</li> <li>Complicated voltage balancing</li> <li>Limited efficiency</li> <li>Complicated structure</li> <li>The higher the capacitors, the lower the reliability</li> <li>Use of more voltage sensors</li> <li>Higher harmonics at low switching frequency rates</li> </ul>
ANPC-MLC	<ul> <li>Balanced distribution of losses</li> <li>Good dynamic response</li> <li>Simplicity in structure</li> <li>Expandability by flying capacitors</li> </ul>	<ul> <li>Higher floating capacitors for levels higher than 3L configuration</li> <li>Complicated control, as opposed to NPC configuration</li> <li>Series switches are needed to operate higher voltage levels</li> </ul>
CHB-MLC	<ul> <li>Modularity in structure</li> <li>Simplicity in structure</li> <li>Good FRT capability</li> <li>Expandability</li> <li>Lack of floating capacitors</li> <li>Straightforward control circuit</li> <li>Low voltage/current harmonics, so no filter is required</li> <li>High reliability</li> </ul>	<ul> <li>Required more switches</li> <li>Inexpensive</li> <li>More DC sources are required</li> <li>Increased volume due to isolated transformers</li> <li>Complicated control circuit</li> <li>Higher number of sensors</li> </ul>
MMC	<ul> <li>Modularity in structure</li> <li>Scalability by cascading similar submodules</li> <li>Simplicity in structure</li> <li>Considerable FRT capability</li> <li>Low losses</li> <li>Minimized modularity level at the submodules</li> <li>Low switching frequency</li> <li>Low quantization on voltage/current (low harmonics)</li> <li>No filter is required</li> <li>High reliability</li> <li>High efficiency</li> </ul>	<ul> <li>Complicated data acquisition and interface between submodules</li> <li>High number of capacitors and submodules, so it can be inexpensive depending on the application</li> <li>Control circuits are required to balance the current and voltage of submodules</li> <li>Higher number of sensors</li> </ul>

Reference [153] presented an impedance-source inverter rather than traditional VSCs for ship systems. The results showed that the proposed configuration offers lower voltage stress on the elements and higher reliability.

Table 4 summarizes the most used MLC topologies and their advantages and disadvantages. Figures 52a, 52b, 52c, 52d, and 52e illustrate the leg schematic of various MLC configurations such as NPC-MLC, FC-MLC, ANPC-MLC, CHB-MLC, and MMC, respectively.

## 4) PEBBS CONCEPT

ONR introduced the concept of PEBB in the mid-1990s to tackle the issues related to power electronics converters, such as cost, availability, design, and operation intricacy. According to [137], the PEBB is a method of building power electronics that uses a standardized approach with modules that can be readily configured to satisfy different applications' specific hardware/software requirements. It includes various blocks such as power processors for single or multiple

phases, gate drives, sensors, signal converters for sensors, switching or gate drive control blocks, and communication blocks for interacting with controllers and with one another.

The PEBB concept seeks to enable dramatic enhancements in the performance, reliability, and cost-effectiveness of electric energy processing systems by developing an approach to the creation and manufacturing of power electronics systems based on the integration of a set of building blocks with (1) integrated functionality, (2) standardized interfaces, (3) suitability for automated manufacturing and mass production, and (4) application versatility [154]. As mentioned in [137], "each PEBB includes built-in self-protection against voltage surges, overvoltage and undervoltage, fault currents, ground currents, internal faults, overloads, and overtemperature." The PEBB concept is designed to include standard interfaces, modularity, and integrated functionalities to enhance power electronics converters' performance, reliability, and cost-effectiveness. AES power converters have power ratings ranging from tens to hundreds of kilowatts at low



FIGURE 52. Leg schematic of different MLC configurations [140], [146].

voltages (<1 kV) to tens of megawatts at medium voltages (1 kV < & <13.8 kV), as shown in Figure 51. The PEBB approach is used in industrial power converters and is well-suited for AES and other ship systems. Using PEBB modules can simplify the development of power converters for various complicated applications.

Despite their high power-conversion efficiency, the heat dissipated by these PEBBs poses serious thermal challenges, meaning that the thermal aspects of the PEBBs concept in AESs deserve more intense scrutiny to complement the improved electrical performance by providing proper cooling in all conceivable scenarios. Traditional PEBB cooling strategies include heat sinks, direct liquid cooling, and heat pipes. Reference [155] presented an in-depth review of traditional and recent electronic cooling techniques applicable to PEBBs. Nonetheless, the design requirements and limitations imposed on the PEBBs concept in AESs, particularly in pursuit of shipboard power corridor [156] to which the PEBBs may be allotted, can place limitations on the application of conventional cooling methods. The power corridor's concept presents extra design constraints like compactness for the PEBBs. Many cooling strategies apply to the shipboard PEBBs, such as PEBB 1000 and PEBB 6000 [157].

A high-power 57.6 kW PEBB with 1.2 kV DC-link and 1.7 kV SiC MOSFET modules is presented in [158] in which the selection of DC-link capacitors is proposed to control the thermal behavior of the PEBBs concept. The results showed that the temperature rise of SiC modules was low. Reference [159] proposed a dynamic electro-thermal approach through an iterative algorithm for the PEBB concept. The proposed approach improved the power loss distribution model, which has temperature-dependent self-improvement ability and an effective thermal model providing a fast temperature calculation.

Depending on failures and the effectiveness of ATC methods, the PEBB stacks are prone to degrade due to aging and environmental stresses. Therefore, degradation is inevitable. To avert this, the lifetime of the PEBB must be assessed so that maintenance works can be planned ahead. Reference [160] proposed a component stress study to estimate the RUL of the PEBB. Reference [161] presented the concept of a standardized switching cell as an additional building block in converter optimization procedures. It is based on modeling a half-bridge arrangement to improve efficiency, volume, and thermal performance.

In [162], the authors tested and simulated 10 kV, 120 A all SiC half-bridge modules for use in a 4 kV, 100 A PEBB. In [163], an Impedance Measurement Unit (IMU) for 1 kVDC/800 VAC systems with an impedance measurement range from 10 Hz to 1 kHz has been proposed. The IMU remarks a reconfigurable, modular, and scalable structure capable of shunting current and series voltage injection mode. The core control of the perturbation injection unit was applied as per a distributed control scheme. The IMU is built with PEBB units and uses 1.7 kV SiC metal oxide semiconductor field effect transistor (MOSFET) modules. Reference [164] describes the development, manufacturing, and preliminary testing of a 1.7 kV SiC switching cell intended for use in a 250 kW integrated PEBB (iPEBB). In [165], a full-bridge PEBB was produced, which includes all ancillary circuits. The individual PEBB is rated for 6 kV operation, but the insulation design of critical systems allows a common mode voltage of up to 30 kV in an MMC. Similarly, a rapid PEBB thermal management tool, vemPEBB, has been proposed in [166] to tailor multi-fidelity PEBB modeling within the setbased design framework. An isolated gate drive power supply for a 5 kV full bridge PEBB consisting of 4 Austin SuperMOS devices has been developed in [167].

PEBBs concept is a practical solution in developing AES power systems with improved reliability, modularity, scalability, FRT capability, lower production, and lower maintenance costs. Figure 53a represents the open plug-and-play architecture of the PEBB concept that can be seamlessly integrated into various applications and automatically configure all its operational settings, eliminating the need for



FIGURE 53. PEBBs concept.

manual adjustments. When the PEBB is plugged into, the host application would immediately detect its presence, specify the producer, and initiate the automatic operation. Each power processing unit within the PEBB would independently control its own parameters for safe operation. The PEBB can be utilized in various applications using standardized interfaces and protocols. These characteristics contribute to cost reduction in defense and commercial sectors [168], [169]. Also, as depicted in Figure 53b, the PEBBs concept offers power converter configurations as modular systems shaped by many submodules connected in series, parallel, or series/parallel [168]. Figure 54 shows some state-of-the-art modular converter configurations based on PEBBs as follows: Series-connected configurations such as CHB for motor drives and MMC for High-Voltage Direct Current (HVDC) systems, parallel-connected configurations such as multi-string PV systems and wind power systems, and series/parallel configuration such as ST systems, and most importantly, AES power systems, as illustrated in Figures 54a, 54b, 54c, 54d, 54e, and 54f, respectively. Plus, Table 5 provides a summary of future opportunities for ATC in different PEBB applications.

#### **B. COMMUNICATION PROTOCOLS OF PEBBS**

One important aspect of ATC in PEBBs concept is the communication protocols with which the data is transferred between PEBBs. This can play an even more critical role in the ATC if the thermal-related data can be communicated faster and more precisely. As discussed previously, the most

efficient type of PSD is the SiC which increases the switching frequency to around 10-20 kHz. Thus, a short cycle time in the microsecond range is vital. This stresses the growing need for a high-speed data transmission communication protocol. Over the years, various communication protocols for power electronics converters have been proposed [171], [172], [173]. The author in [174] established the primary criterion for communication in power electronics converters: simplicity, real-time communication, data rate, high-speed data transmission, data update cycles, synchronization, reliability and availability, redundancy, diagnostics, security, configuration, scalability, and reconfigurability.

PEBBs offer the plug-and-play architecture for various large-scale applications, providing a network with primary/secondary infrastructure. Several communication protocols have been proposed over the years [171], the important ones are controller area network (CAN) [175], serial realtime communications system (SERCOS) [176], motion and control ring optical (MACRO) [177], [178], [179], which is also an Ethernet communication protocol, MACRO and fiber distributed data interface (FDDI) to form power electronics system network (PESNet) [177], [180], PESNet 1.2 and 2.2 as the very two first communication protocols generated PEBBs [181], process field bus (PROFIBUS) [182], serial peripheral interface (SPI) [183], ethernet for control automation technology (EtherCAT) which interestingly most of the researchers have lately zeroed in on because of real-time capabilities, synchronization performance, fault diagnostics [184], process field net ssochronous real time (PROFINET IRT) [184], Ethernet PowerLink and Ethernet/IP [185], synchronous-converter-control-bus (SyCCo-Bus) [186], and RealSycho [187]. A good comparison between PESNet and SyCCo-Bus protocols has been presented in [188].

Some new communication protocols have been recently proposed, such as PESNet 3.0 [189]. In this paper, the comparison of the minimum cycle time between EtherCAT and PESNet 3.0 communication protocols has been drawn, and the results showed that, as opposed to EtherCAT, the innerloop control bandwidth could be boosted in PESNet 3.0. In [190], a distributed communication and control system (DCCS) for high-frequency large-scale modular IMUs with tight synchronization and low latency has been presented for improving PEBBs' modularity and scalability. Reference [191] demonstrated a power converter structure that utilizes data transmission between nodes like gate drivers and sensors, enabling a more flexible and distributed control methodology. Reference [192] proposed PESNet 3.0 for high-power SiC MMCs in which the white rabbit network (WRN) is embedded in PESNet 3.0. For the first time, the WRN has been used in the power electronics in this paper, and the outcomes showed that it could remove the sync error (SE) restrictions of field-programmable gate array (FPGA).



FIGURE 54. State-of-the-art modular converter configurations using PEBBs concept.

 TABLE 5. Future opportunities for ATC in different PEBB applications.

ATC Method	Motor Drives [32]	PV/Wind Systems [32]	STs [32]	AES Power Systems
Gate drive control	***	**	**	-
Switching frequency control	***	**	**	*
Modulation	**	**	**	***
DC-link voltage control	**	*	*	*
Reactive control	*	***	*	*
Active control	-	***	***	*
Profile shaping	-	***	***	**
Power routing	*	***	***	***

Note. \*\*\*: High, \*\*: Medium, \*: Low, -: No.

Moreover, some research has been carried out to provide wireless data transmission for communication [193]. Focusing on the PEBBs concept, [194] tests 5G/6G networks for PEBBs, and the results validated that the delay in data transfer can be enhanced by higher integration of the wireless communication module in the control board.

## C. THERMAL DATA COLLECTION

A major difficulty in the ATC method is deriving thermal state variables, like the junction temperature. To this end, many solutions have recently been put forward that provide higher accuracy and broader bandwidth than traditional methods [195], [196], [197], [198], [199], [200]. Although

some of these solutions can address the issue to some extent, they are mostly inexpensive to implement and lack high signal-to-noise ratios and low sensitivity to external factors [200] and degradation [201]. Furthermore, temperature sensing is inadequate per se to have all the capabilities ATC can offer. The reason why is that the desired filter brings about phase lag. Although thermal predictors offer an affordable, locally resolved, and zero phase lag option for determining temperature, the model errors restrict their precision for complex calibration and degradation, which are not considered. Therefore, thermal observers graft the advantages of both solutions into one another by incorporating temperature sensor data into a closed-loop control system to reduce real-time model errors of the thermal estimator, as described in various studies including [27], [202], and [203].

## D. STATE-OF-HEALTH (SOH) AND DEGRADATION

Many works have focused on degradation forecasting of distribution management systems, which include different types of EVs and batteries, to improve overall reliability by reducing the degradation rate of each component [204], [205], [206]. The operating condition of the components impacts the degradation process [207]. The degradation of the components results in decreased capacity, reducing the system's performance and increasing the reliability concern [208], [209]. The same concept of degradation analysis will be invaluable in power electronics converters since accurately predicting the SoH and the progression of damage is crucial for implementing thermal control in power converters for maintenance purposes. It is important to note that without sufficient data on the components' RULs, it is difficult or even incorrect to use power routing, which could ultimately reduce the system's reliability.

To create the modules' lifetime models, the thermal cycle's data, such as its damage, is gathered linearly [210], [211]. On the one hand, researchers have recently found that aging is non-linear. Therefore, they conducted experiments comparing newly manufactured power modules with pre-aged ones. The findings suggested that the new modules did not experience aging under identical thermal stress. On the other hand, the pre-aged modules showed a continued rise in thermal resistance related to aging. Based on the outcomes, a physics-based lifetime model was created [212]. It should be mentioned that the data creation for the feature of these lifetime models is not straightforward. Real-time power cycling data is essential for temperature-based lifetime models closely connected to various failure mechanisms. The models can only be generated by tracking plastic strain, which is a cause of damage to the modules, throughout a specific operating time. The research to obtain strain from modules needs further development [213], [214].

## E. COST ANALYSIS

This technology raises significant concerns regarding the reliability of power electronics and the associated maintenance costs. The costs can be minimized by monitoring and controlling the lifespan of PSDs [263], [264]. However, addressing these challenges is not a simple task, as it requires understanding power device physics, effectively managing electrical variables, and developing optimal plans and schedules for maintenance.

To the best of the authors' knowledge, no comprehensive analysis has been conducted on the cost of power converters when employing ATC methods. While the literature emphasizes the cost reduction associated with these methods, no specific cost analysis or numerical comparison has been performed. An improved ATC method offers several advantages. Firstly, it enables the use of narrower safety margins, leading to enhanced reliability, increased availability, and, usually, cooler systems. Secondly, it reduces the expenses incurred from unscheduled maintenance. Unscheduled maintenance can result in lost revenue and system unavailability, which become critical factors as operational demands increase and financial resources are limited. Therefore, developing a cost-effective ATC method that considers the SoH and degradation of power converters is crucial to optimize maintenance costs [84].

From a cost reduction perspective in the context of the PEBB concept, adjusting the ATC of converter devices based on the thermal behavior of PEBB stacks is advantageous. This is because if a single PEBB fails, only a direct replacement is required to maintain proper power system operation. Implementing ATC and RUL methods facilitates this capability, allowing PEBBs to adapt their operation to align with scheduled maintenance, thereby increasing the cost-efficiency of planned maintenance tasks. Essentially, designing an effective ATC method with SoH and maintenance schedule will allow replacing system components when their failure probability reaches the maximum limit. This modification of ATC with SoH and RUL estimation will guarantee that the failure is always detected, simplifying maintenance scheduling and reducing associated costs [170].

In recent works [265] and [266], a comprehensive review was presented, highlighting the latest developments and emerging trends in technologies and approaches aimed at improving the dependability of power electronics systems. The authors discussed advancements across various levels, encompassing power electronics topology, control strategies, condition monitoring techniques, and digital-twin systems. They compared various FRT methods for full-bridge isolated DC/DC converters, including cost and number of elements.

## F. REDUCING THE EFFECT ON THE OPERATION OF THE SYSTEM

The use of ATC methods offers an additional capability for PSDs and allows for altering the PWM frequency or system process to protect the PSDs from overtemperature damage. Although these methods have a negligible impact on system function under regular conditions, they can significantly

Criteria	Approaches	Future Work Perspectives
Enhancement	- Decrease in power loss:	• Improving the reliability of oper-
in overload	• Frequency ([26], [27], [117], [215], [216], [217], [218]),	ation close to thermal restrictions
feature	• Peak current ([26], [27], [117], [219], [216], [220]),	• Achieving a more reasonable dif-
	• Duty ratio ([215]),	ference between the maximum
	• Adjust current restriction to coolant temperature and frequency (	junction temperature throughout
	[219]).	the operation and the nominal
	- Increase in power loss:	junction temperature
	• Reduction in temperature swings ([221], [222]).	• Enhancing the operation perfor-
		mance during transient at ther-
		mal restrictions
Decrease in	Manipulation in terms of the following:	• Lowering thermal cycles which
thermal cycle	- Switching power loss:	bring about damages to PSDs
	• Gate registance $([210], [223], [224])$	- Enhancing the longevity of power
	• Gate resistance $([210], [220], [224]),$ • Frequency $([27], [28], [46], [216], [210], [225], [226], [227], [228],$	Dimanening the longevity of power
	• Frequency $([27], [20], [40], [40], [210], [210], [220], [220], [227], [220], [220], [2$	Coming up with an optimum
	(223), • Cate drivers ([22] [220] [222] [221] [222] [222] [224] [225])	• Coming up with an optimum
	• Gate drivers ( $[52]$ , $[250]$ , $[222]$ , $[251]$ , $[252]$ , $[253]$ , $[254]$ , $[255]$ ),	arailability,
	• Shoot-through duty ratio ([255]).	availability, the metime of power
	- Doading:	converters, and F5Ds rating
	• Reactive power now $([31])$ ,	
	• Adjustment of current or voltage restriction ([32], [216], [225]),	
	• Flux control of electric machines (minimized converter's loss) (	
	[230]),	
	• Optimized Maximum Power Point Tracking (MPPT) ([237]),	
	• Optimized current for low-load applications ([238]).	
	- Interface:	
	• Cooling fan ( $[239]$ ),	
	• Liquid metal pump control ([240]),	
	• Thermo-electric cooling control ([241]).	
	- Modulation strategy control:	
	• DPWM ([28], [32], [227], [242], [243]),	
	• FCS-MPC ( [51], [52], [244], [245], [246], [247], [248]).	
Controlling	- Parallel PSDs:	• Keeping an optimal distribution
thermal stress	• Multi-chip modules ( [249], [250], [251], [252], [253]),	of thermal stress across PSDs
across PSDs	• Parallel power modules ([74], [75]).	• Monitoring the degradation and
	- Configurations:	SoH of PSDs in ATC
	• Hybrid ANPC converter ( [248]).	
Controlling	- Series PEBBs:	• Regulating the distribution of
thermal stress	• MMCs ( [69], [71], [254], [255], [256], [257]),	loading on various elements
across PSDs in	• CHB converters ( [93]).	Regulating RUL
PEBBs	- Parallel PEBBs:	• Attaining optimal maintenance
	• Parallel ([32], [242], [258]),	schedules
	• Load sharing in microgrids ( [259]).	
	- Control of certain elements:	
	• Power routing ([54], [56], [57], [77], [84], [89], [258], [260], [261]).	
	- Redundancy concept:	
	• Multi-phase power routing ([78], [105], [262]).	
	- man hunde house reasons ( [.e], [ree]).	

#### TABLE 6. Some important classifications and future outlooks to help ATC of PEBBs concept in AES power systems.

affect system operation when thermal limits are approached. All conditions must be considered to mitigate this effect, but protection against thermal damage stays reliable. The work presented in [219] is a good example. In this study, the maximum peak current of a traction inverter could be doubled to around 200% based on the operating point and not surpassing the maximum junction temperature. This was due to the decaying thermal impedance frequency response as the frequency went up.

## G. PSD TECHNOLOGY AND PACKAGING

ATC methods are becoming increasingly vital in new PSD technologies and packaging. However, it also presents new challenges. WBG semiconductors such as SiC and gallium nitride (GaN) have smaller footprints and thicknesses than

VOLUME 11, 2023

their silicon (Si) equivalents while keeping the same dielectric strength [267], [268]. This makes it possible to create more compact power electronic converters [269]. However, due to the smaller size of the PSDs, their thermal capacitance is lower, resulting in increased thermal/mechanical stress due to higher thermal cycling. This means that the lifetime of WBG-based power modules is more affected by thermal cycles than Si PSDs. Passive methods have been proposed to reduce the thermal cycling of WBG devices by using thermal buffers to increase adequate thermal capacity without degrading the thermal path [270]. As the method brings about a more complicated module packaging, the ATC is an effective method for the lifetime extension of WBG semiconductors [230], [271], [272], [273].

Although ATC using WBG devices has several advantages, there are significant challenges in obtaining accurate measurement and high-bandwidth junction temperature data. With WBG semiconductors, various temperature-sensitive electrical parameters (TSEPs) that can be obtained from high bandwidth are not applicable in the same way as with Si devices or require a more intricate circuit for data collection. This issue has been discussed in several research papers, including [197], [198], [201], [274], and [275].

The ever-growing use of WBG semiconductors has revolutionized module packaging to improve the deployment of enhanced electrical features, unlike Si devices. Bond-wirefree transfer-molded power modules with lead frames are used to increase integration density, which can withstand higher thermal stress than soft silicone encapsulation modules. However, such modules still experience thermal cyclinginduced aging effects, such as delamination of the chip from the lead frame and mold cracking [276], [277]. Recent solutions, such as board integration of PSDs, have been presented to decrease switch cell inductance and enhance system integration. Still, they can degrade the semiconductors' thermal path, leading to increased thermal cycling [278]. Hence, using ATC methods dramatically improves the reliability and lifetime of future packaging of WBGs.

Table 6 summarizes some hot research spots and radical challenges for future work.

#### **V. CONCLUSION**

The ATC method is used to improve the performance and reliability of a power converter by regulating its thermal behavior through control techniques. The main objective of the ATC techniques is to lower the average temperature and swing amplitude in PSDs. The main highlights of this paper are as follows:

- The ATC methods in different converter configurations; such as single configurations (two-level, threelevel, multilevel converters), parallel configurations, cascaded configurations, and modular configurations; have been reviewed, and a detailed comparison has been provided.
- 2) A brief overview of the most commonly used thermal networks and their comparison has been provided. The mathematical and theoretical principles of these networks deserve a separate review paper.
- 3) The ATC methods with the power routing approach have been presented. There are no papers related to power routing for single configurations because it does not apply to them. But many papers propose the optimal power distribution between modular power converters.
- 4) Different layers of thermal control based on various bandwidths have been discussed in detail. The layers include the modulation level (e.g., modulation with switching power loss reduction, modulation with conduction loss redistribution), the system function level (e.g., controlling reactive power, active power, or ESS), and other levels (e.g., active gate control, switching frequency, manipulation of the loss distribution between

IGBTs and diodes). And a detailed comparison of different control layers has been provided.

- 5) Next, the PEBBs concept has been introduced and discussed. So far, various ATC methods have been presented based on control levels, configurations, and modulations. After that, the discussion continued with the PEBBs concept and the most important converter configurations to form PEBBs stacks, such as NPC-MLC, FC-MLC, ANPC-MLC, CHB-MLC, and the most important one, MMC. The MMC configuration is the best candidate for the PEBBs concept to meet AES power systems' requirements, such as modularity, scalability and simplicity in structure, hot-swapping, redundancy, low filter requirements, and lower quantization on current and voltage. The advantages and disadvantages of each one of these configurations in the PEBBs concept have been presented.
- 6) Having several MMCs for shaping PEBB stacks raises thermal stress. Hence, a precise yet cost-effective ATC method is very much needed to control the thermal behavior of the PEBBs concept. Since research has yet to be carried out on thermal control of the PEBBs concept, this paper tried to explain the whys and the wherefores of implementation of the ATC methods discussed in this paper to this concept. Also, the communication protocols that help the ATC methods in this concept have been briefly discussed, which must be improved for better data acquisition. This helps transfer thermal-related data of each PEBB stack, leading to better health monitoring.
- 7) Finally, a very detailed scope of future work is presented with categorized recent papers of different areas where improvements can/must be made in the PEBBs concept. Also, some future work perspectives, such as thermal data collection, state-of-health, degradation forecasting, cost analysis, PSD technology, PSD packaging, etc., have been proposed.

This work was investigated in a timely and comprehensive manner. It will serve as a comprehensive and useful reference for those who are working on controlling the thermal behavior of power converters and how these methods can be employed in different applications such as PEBBs concept in ship power systems, aircraft, motor drives, HVDC, wind power systems, multi-string PV systems, ST systems, etc.

## REFERENCES

- J. Olamaei, S. Ebrahimi, and A. Moghassemi, "Compensation of voltage sag caused by partial shading in grid-connected PV system through the three-level SVM inverter," *Sustain. Energy Technol. Assessments*, vol. 18, pp. 107–118, Dec. 2016, doi: 10.1016/j.seta.2016.10.001.
- [2] A. Emadi, Y. J. Lee, and K. Rajashekara, "Power electronics and motor drives in electric, hybrid electric, and plug-in hybrid electric vehicles," *IEEE Trans. Ind. Electron.*, vol. 55, no. 6, pp. 2237–2245, Jun. 2008, doi: 10.1109/TIE.2008.922768.
- [3] A. Tahjib, H. Tanzin, S. M. I. Rahman, B. S. Islam, and H. Azim, "Development of a solar powered electric vehicle based on tadpole design," in *Proc. 2nd Int. Conf. Robot., Electr. Signal Process. Techn. (ICREST)*, Jan. 2021, pp. 206–210, doi: 10.1109/ICREST51555.2021.9331052.

- [4] A. Moghassemi and S. Padmanaban, "Dynamic voltage restorer (DVR): A comprehensive review of topologies, power converters, control methods, and modified configurations," *Energies*, vol. 13, no. 16, p. 4152, Aug. 2020, doi: 10.3390/en13164152.
- [5] S. Castellan, R. Menis, A. Tessarolo, F. Luise, and T. Mazzuca, "A review of power electronics equipment for all-electric ship MVDC power systems," *Int. J. Electr. Power Energy Syst.*, vol. 96, pp. 306–323, Mar. 2018, doi: 10.1016/j.ijepes.2017.09.040.
- [6] H. Wang, M. Liserre, and F. Blaabjerg, "Toward reliable power electronics: Challenges, design tools, and opportunities," *IEEE Ind. Electron. Mag.*, vol. 7, no. 2, pp. 17–26, Jun. 2013, doi: 10.1109/MIE.2013.2252958.
- [7] A. Moghassemi, S. Ebrahimi, and F. Ferdowsi, "A novel control scheme for TransZSI-DVR to enhance power quality in solar integrated networks," in *Proc. North Amer. Power Symp. (NAPS)*, Nov. 2021, pp. 1–6, doi: 10.1109/NAPS52732.2021.9654507.
- [8] A. Moghassemi, S. Ebrahimi, S. Padmanaban, M. Mitolo, and J. B. Holm-Nielsen, "Two fast metaheuristic-based MPPT techniques for partially shaded photovoltaic system," *Int. J. Electr. Power Energy Syst.*, vol. 137, May 2022, Art. no. 107567, doi: 10.1016/j.ijepes.2021.107567.
- [9] Application Manual Power Semiconductors, SEMIKRON Int. GmbH, Nuremberg, Germany, 2010.
- [10] S. Yang, A. Bryant, P. Mawby, D. Xiang, L. Ran, and P. Tavner, "An industry-based survey of reliability in power electronic converters," *IEEE Trans. Ind. Appl.*, vol. 47, no. 3, pp. 1441–1451, May 2011, doi: 10.1109/TIA.2011.2124436.
- [11] E. Wolfgang, "Examples for failures in power electronics systems," ECPE Tutorial Rel. Power Electron. Syst., Nuremberg, Germany, Tech. Rep., 2007, pp. 19–20. [Online]. Available: https://www.ecpe.org/
- [12] C. Busca, R. Teodorescu, F. Blaabjerg, S. Munk-Nielsen, L. Helle, T. Abeyasekera, and P. Rodríguez, "An overview of the reliability prediction related aspects of high power IGBTs in wind power applications," *Microelectron. Rel.*, vol. 51, nos. 9–11, pp. 1903–1907, Sep. 2011, doi: 10.1016/j.microrel.2011.06.053.
- [13] B. Ji, V. Pickert, W. Cao, and B. Zahawi, "In situ diagnostics and prognostics of wire bonding faults in IGBT modules for electric vehicle drives," *IEEE Trans. Power Electron.*, vol. 28, no. 12, pp. 5568–5577, Dec. 2013, doi: 10.1109/TPEL.2013.2251358.
- [14] A. Volke and M. Hornkamp, *IGBT Modules: Technologies, Driver and Application*. Neubiberg, Germany: Infineon Technologies AG, 2017.
- [15] I. Kovačević, U. Drofenik, and J. W. Kolar, "New physical model for lifetime estimation of power modules," in *Proc. Int. Power Electron. Conf. (ECCE ASIA)*, Jun. 2010, pp. 2106–2114, doi: 10.1109/IPEC.2010.5543755.
- [16] V. Smet, F. Forest, J.-J. Huselstein, F. Richardeau, Z. Khatir, S. Lefebvre, and M. Berkani, "Ageing and failure modes of IGBT modules in hightemperature power cycling," *IEEE Trans. Ind. Electron.*, vol. 58, no. 10, pp. 4931–4941, Oct. 2011, doi: 10.1109/TIE.2011.2114313.
- [17] T. Herrmann, M. Feller, J. Lutz, R. Bayerer, and T. Licht, "Power cycling induced failure mechanisms in solder layers," in *Proc. Eur. Conf. Power Electron. Appl.*, 2007, pp. 1–7, doi: 10.1109/EPE.2007. 4417702.
- [18] B. Ji, X. Song, E. Sciberras, W. Cao, Y. Hu, and V. Pickert, "Multiobjective design optimization of IGBT power modules considering power cycling and thermal cycling," *IEEE Trans. Power Electron.*, vol. 30, no. 5, pp. 2493–2504, May 2015, doi: 10.1109/TPEL.2014.2365531.
- [19] X. Wang, A. Castellazzi, and P. Zanchetta, "Regulated cooling for reduced thermal cycling of power devices," in *Proc. 7th Int. Power Electron. Motion Control Conf.*, vol. 1, Jun. 2012, pp. 238–244, doi: 10.1109/IPEMC.2012.6258837.
- [20] A. Yahyaee, A. Bahman, and F. Blaabjerg, "A modification of offset strip fin heatsink with high-performance cooling for IGBT modules," *Appl. Sci.*, vol. 10, no. 3, p. 1112, Feb. 2020, doi: 10.3390/app10031112.
- [21] F. Wang, J. Cao, Z. Ling, Z. Zhang, and X. Fang, "Experimental and simulative investigations on a phase change material nano-emulsionbased liquid cooling thermal management system for a lithium-ion battery pack," *Energy*, vol. 207, Sep. 2020, Art. no. 118215, doi: 10.1016/j.energy.2020.118215.
- [22] C. Qian, A. M. Gheitaghy, J. Fan, H. Tang, B. Sun, H. Ye, and G. Zhang, "Thermal management on IGBT power electronic devices and modules," *IEEE Access*, vol. 6, pp. 12868–12884, 2018, doi: 10.1109/ACCESS.2018.2793300.

- [23] J. Liang, Y. Gan, Y. Li, M. Tan, and J. Wang, "Thermal and electrochemical performance of a serially connected battery module using a heat pipe-based thermal management system under different coolant temperatures," *Energy*, vol. 189, Dec. 2019, Art. no. 116233, doi: 10.1016/j.energy.2019.116233.
- [24] H. M. Ali, "Recent advancements in PV cooling and efficiency enhancement integrating phase change materials based systems—A comprehensive review," *Sol. Energy*, vol. 197, pp. 163–198, Feb. 2020, doi: 10.1016/j.solener.2019.11.075.
- [25] Y. Yoon, S. Hyeon, D. R. Kim, and K.-S. Lee, "Minimizing thermal interference effects of multiple heat sources for effective cooling of power conversion electronics," *Energy Convers. Manage.*, vol. 174, pp. 218–226, Oct. 2018, doi: 10.1016/j.enconman.2018.08.047.
- [26] J. Lemmens, P. Vanassche, and J. Driesen, "Optimal control of traction motor drives under electrothermal constraints," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 2, no. 2, pp. 249–263, Jun. 2014, doi: 10.1109/JESTPE.2014.2299765.
- [27] D. A. Murdock, J. E. R. Torres, J. J. Connors, and R. D. Lorenz, "Active thermal control of power electronic modules," *IEEE Trans. Ind. Appl.*, vol. 42, no. 2, pp. 552–558, Mar. 2006, doi: 10.1109/TIA.2005. 863905.
- [28] M. Weckert and J. Roth-Stielow, "Lifetime as a control variable in power electronic systems," in *Proc. Emobility-Elect. Power Train*, 2010, pp. 1–6, doi: 10.1109/EMOBILITY.2010.5668049.
- [29] J. Lemmens, J. Driesen, and P. Vanassche, "Dynamic DC-link voltage adaptation for thermal management of traction drives," in *Proc. IEEE Energy Convers. Congr. Expo.*, Sep. 2013, pp. 180–187, doi: 10.1109/ECCE.2013.6646698.
- [30] C. Sintamarean, H. Wang, F. Blaabjerg, and F. Iannuzzo, "The impact of gate-driver parameters variation and device degradation in the PVinverter lifetime," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2014, pp. 2257–2264, doi: 10.1109/ECCE.2014.6953704.
- [31] K. Ma, M. Liserre, and F. Blaabjerg, "Reactive power influence on the thermal cycling of multi-MW wind power inverter," *IEEE Trans. Ind. Appl.*, vol. 49, no. 2, pp. 922–930, Mar. 2013, doi: 10.1109/TIA.2013.2240644.
- [32] M. Andresen, K. Ma, G. Buticchi, J. Falck, F. Blaabjerg, and M. Liserre, "Junction temperature control for more reliable power electronics," *IEEE Trans. Power Electron.*, vol. 33, no. 1, pp. 765–776, Jan. 2018, doi: 10.1109/TPEL.2017.2665697.
- [33] Transient Thermal Measurements and Thermal Equivalent Circuit Models, Infineon Technologies AG, Neubiberg, Germany, 2020.
- [34] S. Zheng, X. Du, J. Zhang, Y. Yu, Q. Luo, and W. Lu, "Monitoring the thermal grease degradation based on the IGBT junction temperature cooling curves," in *Proc. IEEE Int. Power Electron. Appl. Conf. Expo.* (*PEAC*), Nov. 2018, pp. 1–4, doi: 10.1109/PEAC.2018.8590329.
- [35] U. Drofenik and J. W. Kolar, "Teaching thermal design of power electronic systems with web-based interactive educational software," in *Proc. 18th Annu. IEEE Appl. Power Electron. Conf. Expo.* (APEC), vol. 2, Feb. 2003, pp. 1029–1036, doi: 10.1109/APEC.2003. 1179343.
- [36] A. Kundu, P. Korta, L. V. Iyer, and N. C. Kar, "Power module thermal characterization considering aging towards online state-ofhealth monitoring," in *Proc. IEEE Appl. Power Electron. Conf. Expo.* (APEC), Mar. 2023, pp. 1128–1134, doi: 10.1109/APEC43580.2023. 10131530.
- [37] M. Ciappa, W. Fichtner, T. Kojima, Y. Yamada, and Y. Nishibe, "Extraction of accurate thermal compact models for fast electro-thermal simulation of IGBT modules in hybrid electric vehicles," *Microelectron. Rel.*, vol. 45, nos. 9–11, pp. 1694–1699, Sep. 2005, doi: 10.1016/j.microrel.2005.07.083.
- [38] M. Musallam and C. M. Johnson, "Real-time compact thermal models for health management of power electronics," *IEEE Trans. Power Electron.*, vol. 25, no. 6, pp. 1416–1425, Jun. 2010, doi: 10.1109/TPEL.2010.2040634.
- [39] A. T. Bryant, P. A. Mawby, P. R. Palmer, E. Santi, and J. L. Hudgins, "Exploration of power device reliability using compact device models and fast electrothermal simulation," *IEEE Trans. Ind. Appl.*, vol. 44, no. 3, pp. 894–903, May/Jun. 2008, doi: 10.1109/TIA.2008.921388.
- [40] L. R. G. Reddy, "Lifetime estimation of IGBTs in a grid-connected STATCOM," Ph.D. dissertation, Univ. Tennessee, Knoxville, TN, USA, 2014. [Online]. Available: https://trace.tennessee.edu/utk\_graddiss/3130/

- [41] A. Moghassemi, D. S. Vanaja, J. Olamaei, G. Ozkan, and C. S. Edrington, "A novel switching method in PV-fed quasi-ZSI-DVR for voltage quality enhancement of photovoltaic integrated networks," *IET Renew. Power Gener.*, Aug. 2022, doi: 10.1049/rpg2.12575.
- [42] A. Moghassemi, S. Padmanaban, V. K. Ramachandaramurthy, M. Mitolo, and M. Benbouzid, "A novel solar photovoltaic fed TransZSI-DVR for power quality improvement of grid-connected PV systems," *IEEE Access*, vol. 9, pp. 7263–7279, 2021, doi: 10.1109/ACCESS.2020.3048022.
- [43] C. Phanikumar, J. Roy, and V. Agarwal, "A hybrid nine-level, 1-φ grid connected multilevel inverter with low switch count and innovative voltage regulation techniques across auxiliary capacitor," *IEEE Trans. Power Electron.*, vol. 34, no. 3, pp. 2159–2170, Mar. 2019, doi: 10.1109/TPEL.2018.2846628.
- [44] P. Chamarthi, P. Chhetri, and V. Agarwal, "Simplified implementation scheme for space vector pulse width modulation of n-level inverter with online computation of optimal switching pulse durations," *IEEE Trans. Ind. Electron.*, vol. 63, no. 11, pp. 6695–6704, Nov. 2016, doi: 10.1109/TIE.2016.2586438.
- [45] M. Andresen, G. Buticchi, J. Falck, M. Liserre, and O. Mühlfeld, "Active thermal management for a single-phase H-bridge inverter employing switching frequency control," in *Proc. Int. Exhib. Conf. Power Electron.*, *Intell. Motion, Renew. Energy Energy Manage. (PCIM Eur.)*, May 2015, pp. 1–8.
- [46] L. Wei, J. McGuire, and R. A. Lukaszewski, "Analysis of PWM frequency control to improve the lifetime of PWM inverter," *IEEE Trans. Ind. Appl.*, vol. 47, no. 2, pp. 922–929, Mar. 2011, doi: 10.1109/TIA.2010.2103391.
- [47] T.-M. Phan, N. Oikonomou, G. J. Riedel, and M. Pacas, "PWM for active thermal protection in three level neutral point clamped inverters," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2014, pp. 3710–3716, doi: 10.1109/ECCE.2014.6953905.
- [48] K. Ma and F. Blaabjerg, "Modulation methods for neutral-point-clamped wind power converter achieving loss and thermal redistribution under low-voltage ride-through," *IEEE Trans. Ind. Electron.*, vol. 61, no. 2, pp. 835–845, Feb. 2014, doi: 10.1109/TIE.2013.2254099.
- [49] K. Ma and F. Blaabjerg, "Thermal optimised modulation methods of three-level neutral-point-clamped inverter for 10 MW wind turbines under low-voltage ride through," *IET Power Electron.*, vol. 5, no. 6, pp. 920–927, Jul. 2012, doi: 10.1049/iet-pel.2011.0446.
- [50] E. Buraimoh, I. E. Davidson, and F. Martinez-Rodrigo, "Fault ridethrough enhancement of grid supporting inverter-based microgrid using delayed signal cancellation algorithm secondary control," *Energies*, vol. 12, no. 20, p. 3994, Oct. 2019, doi: 10.3390/en12203994.
- [51] J. Falck, G. Buticchi, and M. Liserre, "Thermal stress based model predictive control of electric drives," *IEEE Trans. Ind. Appl.*, vol. 54, no. 2, pp. 1513–1522, Mar. 2018, doi: 10.1109/TIA.2017. 2772198.
- [52] G. Ozkan, B. Papari, P. H. Hoang, N. Deb, and C. S. Edrington, "An active thermal control method for AC-DC power converter with sequence-based control approach," in *Proc. IEEE Electr. Ship Technol. Symp. (ESTS)*, Aug. 2019, pp. 263–267, doi: 10.1109/ESTS.2019. 8847886.
- [53] D. Ronanki and S. S. Williamson, "Modular multilevel converters for transportation electrification: Challenges and opportunities," *IEEE Trans. Transport. Electrific.*, vol. 4, no. 2, pp. 399–407, Jun. 2018, doi: 10.1109/TTE.2018.2792330.
- [54] M. Liserre, M. Andresen, L. Costa, and G. Buticchi, "Power routing in modular smart transformers: Active thermal control through uneven loading of cells," *IEEE Ind. Electron. Mag.*, vol. 10, no. 3, pp. 43–53, Sep. 2016, doi: 10.1109/MIE.2016.2588898.
- [55] Y. Ko, M. Andresen, G. Buticchi, and M. Liserre, "Power routing for cascaded H-bridge converters," *IEEE Trans. Power Electron.*, vol. 32, no. 12, pp. 9435–9446, Dec. 2017, doi: 10.1109/TPEL.2017. 2658182.
- [56] M. Andresen, V. Raveendran, G. Buticchi, and M. Liserre, "Lifetimebased power routing in parallel converters for smart transformer application," *IEEE Trans. Ind. Electron.*, vol. 65, no. 2, pp. 1675–1684, Feb. 2018, doi: 10.1109/TIE.2017.2733426.
- [57] V. Raveendran, M. Andresen, G. Buticchi, and M. Liserre, "Thermal stress based power routing of smart transformer with CHB and DAB converters," *IEEE Trans. Power Electron.*, vol. 35, no. 4, pp. 4205–4215, Apr. 2020, doi: 10.1109/TPEL.2019.2935249.

- [58] J. Li, "Design and control optimisation of a novel bypass-embedded multilevel multicell inverter for hybrid electric vehicle drives," in *Proc. IEEE 11th Int. Symp. Power Electron. Distrib. Gener. Syst. (PEDG)*, Sep. 2020, pp. 382–385, doi: 10.1109/PEDG48541.2020.9244313.
- [59] J. Wang, X. Liu, Q. Xiao, D. Zhou, H. Qiu, and Y. Tang, "Modulated model predictive control for modular multilevel converters with easy implementation and enhanced steady-state performance," *IEEE Trans. Power Electron.*, vol. 35, no. 9, pp. 9107–9118, Sep. 2020, doi: 10.1109/TPEL.2020.2969688.
- [60] M. H. Nguyen, S. Kwak, and T. Kim, "Phase-shifted carrier pulse-width modulation algorithm with improved dynamic performance for modular multilevel converters," *IEEE Access*, vol. 7, pp. 170949–170960, 2019, doi: 10.1109/ACCESS.2019.2955714.
- [61] M. H. Nguyen and S. Kwak, "Nearest-level control method with improved output quality for modular multilevel converters," *IEEE Access*, vol. 8, pp. 110237–110250, 2020, doi: 10.1109/ACCESS.2020.3001587.
- [62] B. Gutierrez and S.-S. Kwak, "Modular multilevel converters (MMCs) controlled by model predictive control with reduced calculation burden," *IEEE Trans. Power Electron.*, vol. 33, no. 11, pp. 9176–9187, Nov. 2018, doi: 10.1109/TPEL.2018.2789455.
- [63] D. Ronanki and S. S. Williamson, "A novel 2N+1 carrier-based pulse width modulation scheme for modular multilevel converters with reduced control complexity," *IEEE Trans. Ind. Appl.*, vol. 56, no. 5, pp. 5593–5602, Sep./Oct. 2020, doi: 10.1109/APEC.2019. 8722205.
- [64] M. H. Nguyen and S. Kwak, "Simplified indirect model predictive control method for a modular multilevel converter," *IEEE Access*, vol. 6, pp. 62405–62418, 2018, doi: 10.1109/ACCESS.2018.2876505.
- [65] Z. Wang, H. Wang, Y. Zhang, and F. Blaabjerg, "Submodule level power loss balancing control for modular multilevel converters," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2018, pp. 5731–5736, doi: 10.1109/ECCE.2018.8557496.
- [66] M. H. Nguyen and S.-S. Kwak, "Switching loss balancing technique for modular multilevel converters operated by model predictive control method," *Electronics*, vol. 8, no. 10, p. 1175, Oct. 2019, doi: 10.3390/electronics8101175.
- [67] A. Sangwongwanich, L. Mathe, R. Teodorescu, C. Lascu, and L. Harnefors, "Two-dimension sorting and selection algorithm featuring thermal balancing control for modular multilevel converters," in *Proc. 18th Eur. Conf. Power Electron. Appl. (EPE ECCE Europe)*, Sep. 2016, pp. 1–10, doi: 10.1109/EPE.2016.7695268.
- [68] M. H. Nguyen and S. Kwak, "Enhance reliability of semiconductor devices in power converters," *Electronics*, vol. 9, no. 12, p. 2068, Dec. 2020, doi: 10.3390/electronics9122068.
- [69] F. Hahn, M. Andresen, G. Buticchi, and M. Liserre, "Thermal analysis and balancing for modular multilevel converters in HVDC applications," *IEEE Trans. Power Electron.*, vol. 33, no. 3, pp. 1985–1996, Mar. 2018, doi: 10.1109/TPEL.2017.2691012.
- [70] J. Gonçalves, D. J. Rogers, and J. Liang, "Submodule temperature regulation and balancing in modular multilevel converters," *IEEE Trans. Ind. Electron.*, vol. 65, no. 9, pp. 7085–7094, Sep. 2018, doi: 10.1109/TIE.2018.2795588.
- [71] J. Sheng, H. Yang, C. Li, M. Chen, W. Li, X. He, and X. Gu, "Active thermal control for hybrid modular multilevel converter under overmodulation operation," *IEEE Trans. Power Electron.*, vol. 35, no. 4, pp. 4242–4255, Apr. 2020, doi: 10.1109/TPEL.2019.2936010.
- [72] M. Hagiwara and H. Akagi, "Control and experiment of pulsewidth-modulated modular multilevel converters," *IEEE Trans. Power Electron.*, vol. 24, no. 7, pp. 1737–1746, Jul. 2009, doi: 10.1109/TPEL.2009.2014236.
- [73] C. Nesgaard and M. A. Andersen, "Optimized load sharing control by means of thermal reliability management," in *Proc. IEEE 35th Annu. Power Electron. Spec. Conf.*, vol. 6, Jun. 2004, pp. 4901–4906, doi: 10.1109/PESC.2004.1354866.
- [74] C. J. Joseph, M. Zolghadri, A. Homaifar, F. Lee, and R. Lorenz, "Novel thermal based current sharing control of parallel converters," in *Proc. 26th Annu. Int. Telecommun. Energy Conf. (INTELEC)*, 2004, pp. 647–653, doi: 10.1109/INTLEC.2004.1401539.
- [75] J. L. Barnette, M. R. Zolghadri, M. Walters, and A. Homaifar, "Temperature integrated load sharing of paralleled modules," in *Proc. 1st IEEE Conf. Ind. Electron. Appl.*, May 2006, pp. 1–6, doi: 10.1109/ICIEA.2006.257370.

- [76] S. Peyghami, P. Davari, and F. Blaabjerg, "System-level lifetime-oriented power sharing control of paralleled DC/DC converters," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2018, pp. 1890–1895, doi: 10.1109/APEC.2018.8341275.
- [77] A. Marquez, J. I. Leon, S. Vazquez, and L. G. Franquelo, "Closedloop active thermal control via power routing of parallel DC–DC converters," in *Proc. IEEE 12th Int. Conf. Compat., Power Electron. Power Eng. (CPE-POWERENG)*, Apr. 2018, pp. 1–6, doi: 10.1109/CPE.2018.8372586.
- [78] H. Yan, G. Buticchi, J. Yang, W. Zhao, H. Zhang, and C. Gerada, "Active thermal control for power converters in modular winding permanent magnet synchronous motor," in *Proc. IEEE 13th Int. Conf. Compat.*, *Power Electron. Power Eng. (CPE-POWERENG)*, Apr. 2019, pp. 1–6, doi: 10.1109/CPE.2019.8862416.
- [79] J. Zhang, G. Chen, and X. Cai, "Thermal smooth control for multi-MW parallel wind power converter," in *Proc. IEEE Int. Conf. IEEE Region (TENCON)*, Oct. 2013, pp. 1–4, doi: 10.1109/TENCON.2013. 6718875.
- [80] J. Zhang, J. Wang, and X. Cai, "Active thermal control-based anticondensation strategy in paralleled wind power converters by adjusting reactive circulating current," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 6, no. 1, pp. 277–291, Mar. 2018, doi: 10.1109/JESTPE.2017. 2741447.
- [81] J. M. Guerrero, J. C. Vasquez, J. Matas, L. G. de Vicuna, and M. Castilla, "Hierarchical control of droop-controlled AC and DC microgrids— A general approach toward standardization," *IEEE Trans. Ind. Electron.*, vol. 58, no. 1, pp. 158–172, Jan. 2011, doi: 10.1109/TIE.2010.2066534.
- [82] S. Peyghami, H. Mokhtari, and F. Blaabjerg, "Decentralized load sharing in a low-voltage direct current microgrid with an adaptive droop approach based on a superimposed frequency," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 5, no. 3, pp. 1205–1215, Sep. 2017, doi: 10.1109/JESTPE.2017.2674300.
- [83] J. Zhang, Y. Li, H. Wang, X. Cai, S. Igarashi, and Z. Wang, "Thermal smooth control based on orthogonal circulating current for multi-MW parallel wind power converter," in *Proc. Int. Power Electron. Appl. Conf. Expo.*, Nov. 2014, pp. 146–151, doi: 10.1109/PEAC.2014.7037844.
- [84] M. Liserre, G. Buticchi, J. I. Leon, A. M. Alcaide, V. Raveendran, Y. Ko, M. Andresen, V. G. Monopoli, and L. Franquelo, "Power routing: A new paradigm for maintenance scheduling," *IEEE Ind. Electron. Mag.*, vol. 14, no. 3, pp. 33–45, Sep. 2020, doi: 10.1109/MIE.2020.2975049.
- [85] M. Andresen, L. F. Costa, G. Buticchi, and M. Liserre, "Smart transformer reliability and efficiency through modularity," in *Proc. IEEE* 8th Int. Power Electron. Motion Control Conf. (IPEMC-ECCE Asia), May 2016, pp. 3241–3248, doi: 10.1109/IPEMC.2016.7512814.
- [86] A. Marquez, J. I. Leon, S. Vazquez, L. G. Franquelo, G. Buticchi, and M. Liserre, "Power device lifetime extension of DC–DC interleaved converters via power routing," in *Proc. 44th Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Oct. 2018, pp. 5332–5337, doi: 10.1109/IECON.2018.8592912.
- [87] V. Raveendran, M. Andresen, and M. Liserre, "Improving onboard converter reliability for more electric aircraft with lifetime-based control," *IEEE Trans. Ind. Electron.*, vol. 66, no. 7, pp. 5787–5796, Jul. 2019, doi: 10.1109/TIE.2018.2889626.
- [88] V. Raveendran, M. Andresen, and M. Liserre, "Reliability oriented control of DC/DC converters for more electric aircraft," in *Proc. IEEE* 27th Int. Symp. Ind. Electron. (ISIE), Jun. 2018, pp. 1352–1358, doi: 10.1109/ISIE.2018.8433853.
- [89] M. Andresen, J. Kuprat, V. Raveendran, J. Falck, and M. Liserre, "Active thermal control for delaying maintenance of power electronics converters," *Chin. J. Electr. Eng.*, vol. 4, no. 3, pp. 13–20, Sep. 2018, doi: 10.23919/CJEE.2018.8471285.
- [90] G. Buticchi, M. Andresen, M. Wutti, and M. Liserre, "Lifetime-based power routing of a quadruple active bridge DC/DC converter," *IEEE Trans. Power Electron.*, vol. 32, no. 11, pp. 8892–8903, Nov. 2017, doi: 10.1109/TPEL.2017.2650258.
- [91] Y. Ko, M. Andresen, G. Buticchi, and M. Liserre, "Thermally compensated discontinuous modulation strategy for cascaded H-bridge converters," *IEEE Trans. Power Electron.*, vol. 33, no. 3, pp. 2704–2713, Mar. 2018, doi: 10.1109/TPEL.2017.2694455.
- [92] Y. Ko, V. Raveendran, M. Andresen, and M. Liserre, "Thermally compensated discontinuous modulation for MVAC/LVDC building blocks of modular smart transformers," *IEEE Trans. Power Electron.*, vol. 35, no. 1, pp. 220–231, Jan. 2020, doi: 10.1109/TPEL.2019.2908853.

- [94] V. G. Monopoli, A. Marquez, J. I. Leon, Y. Ko, G. Buticchi, and M. Liserre, "Improved harmonic performance of cascaded H-bridge converters with thermal control," *IEEE Trans. Ind. Electron.*, vol. 66, no. 7, pp. 4982–4991, Jul. 2019, doi: 10.1109/TIE.2018.2868304.
- [95] A. Albarbar and C. Batunlu, *Thermal Analysis of Power Electronic Devices Used in Renewable Energy Systems*. Cham, Switzerland: Springer, 2018, doi: 10.1007/978-3-319-59828-4.
- [96] J. Mathew and S. Krishnan, "A review on transient thermal management of electronic devices," *J. Electron. Packag.*, vol. 144, no. 1, 2022, Art. no. 010801, doi: 10.1115/1.4050002.
- [97] D. Hirschmann, D. Tissen, S. Schroder, and R. W. De Doncker, "Inverter design for hybrid electrical vehicles considering mission profiles," in *Proc. IEEE Vehicle Power Propuls. Conf.*, Sep. 2005, p. 6, doi: 10.1109/VPPC.2005.1554616.
- [98] J. Due, S. Munk-Nielsen, and R. Nielsen, "Lifetime investigation of high power IGBT modules," in *Proc. 14th Eur. Conf. Power Electron. Appl.*, Aug. 2011, pp. 1–8.
- [99] A. Abuelnaga, M. Narimani, and A. S. Bahman, "Power electronic converter reliability and prognosis review focusing on power switch module failures," *J. Power Electron.*, vol. 21, no. 6, pp. 865–880, Jun. 2021, doi: 10.1007/s43236-021-00228-6.
- [100] Y. Luo, F. Xiao, B. Wang, and B. Liu, "Failure analysis of power electronic devices and their applications under extreme conditions," *Chin. J. Electr. Eng.*, vol. 2, no. 1, pp. 91–100, Jun. 2016, doi: 10.23919/CJEE.2016.7933119.
- [101] J. Kuprat, C. H. van der Broeck, M. Andresen, S. Kalker, M. Liserre, and R. W. De Doncker, "Research on active thermal control: Actual status and future trends," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 9, no. 6, pp. 6494–6506, Dec. 2021, doi: 10.1109/JESTPE.2021.3067782.
- [102] L. Liu, C. Du, Q. Peng, J. Chen, Y. Wang, Y. Chen, Z. Peng, H. Jiang, and L. Ran, "An investigation on IGBT junction temperature estimation using online regression method," *Microelectron. Rel.*, vol. 124, Sep. 2021, Art. no. 114321, doi: 10.1016/j.microrel.2021.114321.
- [103] U. Scheuermann, "Reliability challenges of automotive power electronics," *Microelectron. Rel.*, vol. 49, nos. 9–11, pp. 1319–1325, Sep. 2009, doi: 10.1016/j.microrel.2009.06.045.
- [104] U. Scheuermann and R. Schmidt, "A new lifetime model for advanced power modules with sintered chips and optimized Al wire bonds," in *Proc. Power Convers. Intell. Motion (PCIM Eur.)*, 2013, pp. 810–813.
- [105] H. Yan, W. Zhao, G. Buticchi, and C. Gerada, "Active thermal control for modular power converters in multi-phase permanent magnet synchronous motor drive system," *IEEE Access*, vol. 9, pp. 7054–7063, 2021, doi: 10.1109/ACCESS.2021.3049293.
- [106] K. Ma, Y. Yang, and F. Blaabjerg, "Transient modelling of loss and thermal dynamics in power semiconductor devices," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2014, pp. 5495–5501, doi: 10.1109/ECCE.2014.6954154.
- [107] M. Andresen, G. Buticchi, and M. Liserre, "Study of reliabilityefficiency tradeoff of active thermal control for power electronic systems," *Microelectron. Rel.*, vol. 58, pp. 119–125, Mar. 2016, doi: 10.1016/j.microrel.2015.12.017.
- [108] D. G. Holmes and T. A. Lipo, Pulse Width Modulation for Power Converters: Principles and Practice, vol. 18. Hoboken, NJ, USA: Wiley, 2003.
- [109] A. M. Hava, R. J. Kerkman, and T. A. Lipo, "Carrier-based PWM-VSI overmodulation strategies: Analysis, comparison, and design," *IEEE Trans. Power Electron.*, vol. 13, no. 4, pp. 674–689, Jul. 1998, doi: 10.1109/63.704136.
- [110] Y. Bak and K.-B. Lee, "Discontinuous PWM for low switching losses in indirect matrix converter drives," in *Proc. IEEE Appl. Power Electron. Conf. Exposit. (APEC)*, Mar. 2016, pp. 2764–2769, doi: 10.1109/APEC.2016.7468255.
- [111] Q. Yan, L. Xiao, H. Chen, X. Yuan, H. Xu, and R. Zhao, "An analytical discontinuous space-vector PWM for three-level inverters with unbalanced DC-link voltages," *IEEE Trans. Power Electron.*, vol. 37, no. 7, pp. 7718–7728, Jul. 2022, doi: 10.1109/TPEL.2022.3146277.
- [112] A. Ibrahim and M. Z. Sujod, "Variable switching frequency hybrid PWM technique for switching loss reduction in a three-phase two-level voltage source inverter," *Measurement*, vol. 151, Feb. 2020, Art. no. 107192, doi: 10.1016/j.measurement.2019.107192.

- [113] Y. Wu, M. A. Shafi, A. M. Knight, and R. A. McMahon, "Comparison of the effects of continuous and discontinuous PWM schemes on power losses of voltage-sourced inverters for induction motor drives," *IEEE Trans. Power Electron.*, vol. 26, no. 1, pp. 182–191, Jan. 2011, doi: 10.1109/TPEL.2010.2054837.
- [114] Z. Qin, M. Liserre, and F. Blaabjerg, "Thermal analysis of multi-MW two-level generator side converters with reduced common-mode-voltage modulation methods for wind turbines," in *Proc. 4th IEEE Int. Symp. Power Electron. Distrib. Gener. Syst. (PEDG)*, Jul. 2013, pp. 1–7, doi: 10.1109/PEDG.2013.6785616.
- [115] Z. Zhong, C. Du, X. Xing, C. Qin, Z. Chen, and X. Liu, "An improved DPWM method for single-phase five-level ANPC converters with switching loss reduction and capacitor voltage balance," *IEEE Trans. Transport. Electrific.*, vol. 8, no. 4, pp. 4482–4495, Dec. 2022, doi: 10.1109/TTE.2022.3184954.
- [116] A. Isidori, F. M. Rossi, F. Blaabjerg, and K. Ma, "Thermal loading and reliability of 10-MW multilevel wind power converter at different wind roughness classes," *IEEE Trans. Ind. Appl.*, vol. 50, no. 1, pp. 484–494, Jan. 2014, doi: 10.1109/TIA.2013.2269311.
- [117] V. Blasko, R. Lukaszewski, and R. Sladky, "On line thermal model and thermal management strategy of a three phase voltage source inverter," in *Proc. IEEE Ind. Appl. Conf., 34th IAS Annu. Meeting*, vol. 2, Oct. 1999, pp. 1423–1431, doi: 10.1109/IAS.1999.801687.
- [118] P. K. Chamarthi, M. S. El Moursi, A. Al Durra, K. H. Al Hosani, and A. Al Sumaiti, "Enhanced pulse width modulation methods for 1-φ five-level neutral point clamped inverter," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2022, pp. 1–7, doi: 10.1109/ECCE50734.2022.9947633.
- [119] T. Bruckner and D. G. Holmes, "Optimal pulse-width modulation for three-level inverters," *IEEE Trans. Power Electron.*, vol. 20, no. 1, pp. 82–89, Jan. 2005, doi: 10.1109/tpel.2004.839831.
- [120] K. Ma and F. Blaabjerg, "Modulation methods for three-level neutralpoint-clamped inverter achieving stress redistribution under moderate modulation index," *IEEE Trans. Power Electron.*, vol. 31, no. 1, pp. 5–10, Jan. 2016, doi: 10.1109/TPEL.2015.2451158.
- [121] F. Blaabjerg and K. Ma, "Wind energy systems," *Proc. IEEE*, vol. 105, no. 11, pp. 2116–2131, Nov. 2017, doi: 10.1109/JPROC.2017.2695485.
- [122] F. Blaabjerg and K. Ma, "Future on power electronics for wind turbine systems," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 1, no. 3, pp. 139–152, Sep. 2013, doi: 10.1109/JESTPE.2013.2275978.
- [123] J. Shair, H. Li, J. Hu, and X. Xie, "Power system stability issues, classifications and research prospects in the context of high-penetration of renewables and power electronics," *Renew. Sustain. Energy Rev.*, vol. 145, Jul. 2021, Art. no. 111111, doi: 10.1016/j.rser.2021.111111.
- [124] M. Tsili and S. Papathanassiou, "A review of grid code technical requirements for wind farms," *IET Renew. Power Generat.*, vol. 3, no. 3, pp. 308–332, Sep. 2009, doi: 10.1049/iet-rpg.2008.0070.
- [125] A. Camacho, M. Castilla, J. Miret, R. Guzman, and A. Borrell, "Reactive power control for distributed generation power plants to comply with voltage limits during grid faults," *IEEE Trans. Power Electron.*, vol. 29, no. 11, pp. 6224–6234, Nov. 2014, doi: 10.1109/TPEL.2014.2301463.
- [126] D. Zhou, F. Blaabjerg, M. Lau, and M. Tonnes, "Thermal behavior optimization in multi-MW wind power converter by reactive power circulation," *IEEE Trans. Ind. Appl.*, vol. 50, no. 1, pp. 433–440, Jan. 2014, doi: 10.1109/TIA.2013.2267511.
- [127] S. Engelhardt, I. Erlich, C. Feltes, J. Kretschmann, and F. Shewarega, "Reactive power capability of wind turbines based on doubly fed induction generators," *IEEE Trans. Energy Convers.*, vol. 26, no. 1, pp. 364–372, Mar. 2011, doi: 10.1109/TEC.2010.2081365.
- [128] Z. Qin, M. Liserre, F. Blaabjerg, and H. Wang, "Energy storage system by means of improved thermal performance of a 3 MW grid side wind power converter," in *Proc. 39th Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Nov. 2013, pp. 736–742, doi: 10.1109/IECON.2013.6699226.
- [129] L. Wu and A. Castellazzi, "Temperature adaptive driving of power semiconductor devices," in *Proc. IEEE Int. Symp. Ind. Electron.*, Jul. 2010, pp. 1110–1114, doi: 10.1109/ISIE.2010.5636541.
- [130] X. Wang, Z. Zhao, and L. Yuan, "Current sharing of IGBT modules in parallel with thermal imbalance," in *Proc. IEEE Energy Convers. Congr. Expo.*, Sep. 2010, pp. 2101–2108, doi: 10.1109/ECCE.2010.5618253.
- [131] F. Hahn, G. Buticchi, and M. Liserre, "Active thermal balancing for modular multilevel converters in HVDC applications," in *Proc. 18th Eur. Conf. Power Electron. Appl. (EPE ECCE Eur.)*, Sep. 2016, pp. 1–10, doi: 10.1109/EPE.2016.7695611.

- [132] T. Ericsen, N. Hingorani, and Y. Khersonsky, "Power electronics and future marine electrical systems," *IEEE Trans. Ind. Appl.*, vol. 42, no. 1, pp. 155–163, Jan. 2006, doi: 10.1109/TIA.2005.861306.
- [133] T. Ericsen, "The ship power electronic revolution: Issues and answers," in Proc. 55th IEEE Petroleum Chem. Ind. Tech. Conf., Sep. 2008, pp. 1–11, doi: 10.1109/PCICON.2008.4663984.
- [134] T. Ericsen and A. Tucker, "Power electronics building blocks and potential power modulator applications," in *Proc. 33rd Int. Power Modulator Symp.*, 1998, pp. 12–15, doi: 10.1109/MODSYM.1998.741179.
- [135] T. Ericsen, "Power electronic building blocks—A systematic approach to power electronics," in *Proc. Power Eng. Soc. Summer Meeting*, 2000, pp. 1216–1218, doi: 10.1109/PESS.2000.867553.
- [136] T. Ericsen, N. Hingorani, and Y. Khersonsky, "PEBB—Power electronics building blocks from concept to reality," in *Proc. IEEE Ind. Appl. Soc.* 53rd Annu. Petroleum Chem. Ind. Conf., Sep. 2006, pp. 1–7.
- [137] IEEE Guide for the Design and Application of Power Electronics in Electrical Power Systems on Ships, IEEE Standard 1662-2008, 2009, pp. 1–72, doi: 10.1109/IEEESTD.2009.4804134.
- [138] G. Sulligoi, A. Vicenzutti, and R. Menis, "All-electric ship design: From electrical propulsion to integrated electrical and electronic power systems," *IEEE Trans. Transport. Electrific.*, vol. 2, no. 4, pp. 507–521, Dec. 2016, doi: 10.1109/TTE.2016.2598078.
- [139] K. Satpathi, A. Ukil, and J. Pou, "Short-circuit fault management in DC electric ship propulsion system: Protection requirements, review of existing technologies and future research trends," *IEEE Trans. Transport. Electrific.*, vol. 4, no. 1, pp. 272–291, Mar. 2018, doi: 10.1109/TTE.2017.2788199.
- [140] S. Choudhury, M. Bajaj, T. Dash, S. Kamel, and F. Jurado, "Multilevel inverter: A survey on classical and advanced topologies, control schemes, applications to power system and future prospects," *Energies*, vol. 14, no. 18, p. 5773, Sep. 2021, doi: 10.3390/en14185773.
- [141] P. K. Chamarthi, V. Agarwal, M. S. E. Moursi, and V. Khadkikar, "Novel 1-φ dual input nine-level inverter topology with generalized modulation technique," *IEEE Trans. Energy Convers.*, vol. 37, no. 3, pp. 1789–1802, Sep. 2022, doi: 10.1109/TEC.2022.3140232.
- [142] P. K. Chamarthi, A. Al-Durra, T. H. M. El-Fouly, and K. A. Jaafari, "A novel three-phase transformerless cascaded multilevel inverter topology for grid-connected solar PV applications," *IEEE Trans. Ind. Appl.*, vol. 57, no. 3, pp. 2285–2297, May 2021, doi: 10.1109/TIA.2021.3057312.
- [143] P. Qashqai, A. Sheikholeslami, H. Vahedi, and K. Al-Haddad, "A review on multilevel converter topologies for electric transportation applications," in *Proc. IEEE Vehicle Power Propuls. Conf. (VPPC)*, Oct. 2015, pp. 1–6, doi: 10.1109/VPPC.2015.7352882.
- [144] S. Alepuz, J. Nicolás-Apruzzese, R. Rafiezadeh, S. Busquets-Monge, M. Raya, and A. Filba-Martinez, "Active thermal control in neutral-pointclamped multilevel converters based on switching-cell arrays," 2023, doi: 10.36227/techrxiv.22591789.v1.
- [145] S. Mersche, R. Schreier, P. Himmelmann, and M. Hiller, "Medium voltage power electronic building block for quasi-two-level operation of a flying capacitor converter," in *Proc. 23rd Eur. Conf. Power Electron. Appl. (EPE ECCE Eur.)*, Sep. 2021, pp. P.1–P.10, doi: 10.23919/EPE21ECCEEurope50061.2021.9570590.
- [146] J. Zhang, S. Xu, Z. Din, and X. Hu, "Hybrid multilevel converters: Topologies, evolutions and verifications," *Energies*, vol. 12, no. 4, p. 615, Feb. 2019, doi: 10.3390/en12040615.
- [147] X. Kestelyn and E. Semail, "Multiphase voltage source inverters," in *Power Electronic Converters: PWM Strategies and Current Control Techniques.* New York, NY, USA: Wiley, 2011, ch. 8, doi: 10.1002/9781118621196.ch8.
- [148] E. Levi, R. Bojoi, F. Profumo, H. Toliyat, and S. Williamson, "Multiphase induction motor drives—A technology status review," *IET Electr. Power Appl.*, vol. 1, no. 4, pp. 489–516, 2007, doi: 10.1049/iet-epa:20060342.
- [149] J. S. Thongam, M. Tarbouchi, A. F. Okou, D. Bouchard, and R. Beguenane, "Trends in naval ship propulsion drive motor technology," in *Proc. IEEE Electr. Power Energy Conf.*, Aug. 2013, pp. 1–5, doi: 10.1109/EPEC.2013.6802942.
- [150] J. Li, J. Liu, D. Boroyevich, P. Mattavelli, and Y. Xue, "Three-level active neutral-point-clamped zero-current-transition converter for sustainable energy systems," *IEEE Trans. Power Electron.*, vol. 26, no. 12, pp. 3680–3693, Dec. 2011, doi: 10.1109/TPEL.2011.2161890.
- [151] J. D. Herbst, F. D. Engelkemeir, and A. L. Gattozzi, "High power density and high efficiency converter topologies for electric ships," in *Proc. IEEE Electr. Ship Technol. Symp. (ESTS)*, Apr. 2013, pp. 360–365, doi: 10.1109/ESTS.2013.6523761.

- [152] F. D. Engelkemeir, A. L. Gattozzi, J. D. Herbst, R. E. Hebner, and G. A. Hallock, "Experimental investigation of the performance of two new types of soft-switching power converters for electric ships," in *Proc. IEEE Electr. Ship Technol. Symp. (ESTS)*, Jun. 2015, pp. 261–267, doi: 10.1109/ESTS.2015.7157901.
- [153] Z. Nie, "The development of new-style photovoltaic Z-source inverter of ship," in *Proc. Power Eng. Autom. Conf.*, Sep. 2012, pp. 1–5, doi: 10.1109/PEAM.2012.6612421.
- [154] D. Boroyevich, "Building block integration in power electronics," in *Proc. IEEE Int. Symp. Ind. Electron.*, Jul. 2010, pp. 3673–3678, doi: 10.1109/ISIE.2010.5636059.
- [155] S. M. S. Murshed and C. A. N. de Castro, "A critical review of traditional and emerging techniques and fluids for electronics cooling," *Renew. Sustain. Energy Rev.*, vol. 78, pp. 821–833, Oct. 2017, doi: 10.1016/j.rser.2017.04.112.
- [156] C. M. Cooke, C. Chryssostomidis, and J. Chalfant, "Modular integrated power corridor," in *Proc. IEEE Electric Ship Technol. Symp. (ESTS)*, Aug. 2017, pp. 91–95, doi: 10.1109/ESTS.2017.8069265.
- [157] S. Yang, J. S. Chalfant, J. C. Ordonez, J. A. Khan, C. Li, I. Cvetkovic, J. V. C. Vargas, M. B. Chagas, Y. Xu, R. P. Burgos, and D. Boroyevich, "Shipboard PEBB cooling strategies," in *Proc. IEEE Electr. Ship Technol. Symp. (ESTS)*, Aug. 2019, pp. 24–31, doi: 10.1109/ESTS.2019.8847810.
- [158] J. Wen, H. Jin, X. Dong, C. Wan, Z. Xu, Y. Gan, H. Zhang, and L. Wang, "Design of a high-power power electronics building block based on SiC MOSFET modules," in *Proc. 5th Asia Conf. Energy Electr. Eng. (ACEEE)*, Jul. 2022, pp. 107–112, doi: 10.1109/ACEEE56193.2022.9851863.
- [159] H. Wang, A. M. Khambadkone, and X. Yu, "Dynamic electro-thermal modeling in power electronics building block (PEBB) applications," in *Proc. IEEE Energy Convers. Congr. Expo.*, Sep. 2010, pp. 2993–3000, doi: 10.1109/ECCE.2010.5618360.
- [160] C. L. Toh, Y. M. Wong, and L. E. Norum, "Power electronics building block (PEBB) hardware design and reliability prediction," in *Proc. IEEE Int. Conf. Power Energy (PECon)*, Nov. 2016, pp. 166–171, doi: 10.1109/PECON.2016.7951553.
- [161] G. Papadopoulos and J. Biela, "Standardised switching cell building block for converter design optimisation with detailed electro-thermal model," in *Proc. 24th Eur. Conf. Power Electron. Appl. (EPE ECCE Eur.)*, Sep. 2022, pp. P.1–P.11.
- [162] C. DiMarino, I. Cvetkovic, Z. Shen, R. Burgos, and D. Boroyevich, "10 kV, 120 a SiC MOSFET modules for a power electronics building block (PEBB)," in *Proc. IEEE Workshop Wide Bandgap Power Devices Appl.*, Oct. 2014, pp. 55–58, doi: 10.1109/WiPDA.2014. 6964623.
- [163] S. Zhou, B. Wen, Y. Rong, V. Mitrovic, R. Burgos, J. Verhulst, M. Belkhayat, and D. Boroyevich, "A reconfigurable, modular and scalable impedance measurement unit with SiC MOSFET-based power electronics building blocks," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Jun. 2021, pp. 1047–1054, doi: 10.1109/APEC42165.2021.9487098.
- [164] N. Rajagopal, C. DiMarino, R. Burgos, T. Moaz, I. Cvetkovic, D. Boroyevich, and O. Mathieu, "Design, fabrication, and testing of a 1.7 kV SiC switching cell for a high-density integrated power electronics building block (iPEBB)," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2021, pp. 5247–5254, doi: 10.1109/ECCE47101.2021.9595520.
- [165] J. Stewart, J. Motwani, J. Yu, I. Cvetkovic, and R. Burgos, "Improved power density of a 6 kV, 1 MW power electronics building block through insulation coordination," in *Proc. IEEE 23rd Workshop Control Modeling Power Electron. (COMPEL)*, Jun. 2022, pp. 1–7, doi: 10.1109/COM-PEL53829.2022.9829968.
- [166] S. Yang, J. C. Ordonez, Y. Xu, and I. Cvetkovic, "VemPEBB: Rapid PEBB thermal management tool," in *Proc. IEEE Electr. Ship Technol. Symp. (ESTS)*, Aug. 2021, pp. 1–6, doi: 10.1109/ESTS49166.2021.9512319.
- [167] S. Sen, L. Zhang, X. Feng, and A. Q. Huang, "High isolation auxiliary power supply for medium-voltage power electronics building block," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2020, pp. 2249–2253, doi: 10.1109/APEC39645.2020.9124543.
- [168] T. Ericsen, "The second electronic revolution' (it's all about control)," in *Proc. Ind. Appl. Soc. 56th Annu. Petroleum Chem. Ind. Conf.*, 2009, pp. 1–10, doi: 10.1109/PCICON.2009.5297144.

- [169] B. J. Baliga, "IGBT applications: Defense," in *The IGBT Device: Physics, Design and Applications of the Insulated Gate Bipolar Transistor.* Amsterdam, The Netherlands: Elsevier, 2015, ch. 14, pp. 451–491, doi: 10.1016/B978-1-4557-3143-5.00014-6.
- [170] V. G. Monopoli, A. Marquez, J. I. Leon, M. Liserre, G. Buticchi, L. G. Franquelo, and S. Vazquez, "Applications and modulation methods for modular converters enabling unequal cell power sharing: Carrier variable-angle phase-displacement modulation methods," *IEEE Ind. Electron. Mag.*, vol. 16, no. 1, pp. 19–30, Mar. 2022, doi: 10.1109/MIE.2021.3080232.
- [171] D. Vaidya, S. Mukherjee, M. A. Zagrodnik, and P. Wang, "A review of communication protocols and topologies for power converters," in *Proc. 43rd Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Oct. 2017, pp. 2233–2238, doi: 10.1109/IECON.2017.8216376.
- [172] C. L. Toh and L. E. Norum, "Synchronization mechanisms for internal monitoring and control in power electronics converter," *J. Elect. Eng.*, vol. 14, no. 3, p. 8, 2014.
- [173] C. L. Toh, "Communication network for internal monitoring and control in multilevel power electronics converter," Norwegian Univ. Sci. Technol. (NTNU), Norway, Tech. Rep., 2014.
- [174] Z. Cucej, "Power electronic building blocks: A survey," Informacije MIDEM, vol. 31, no. 2, pp. 94–101.
- [175] Y.-M. Park, H.-S. Yoo, H.-W. Lee, M.-G. Jung, S.-H. Lee, C.-D. Lee, S.-B. Lee, and J.-Y. Yoo, "A simple and reliable PWM synchronization & phase-shift method for cascaded H-bridge multilevel inverters based on a standard serial communication protocol," in *Proc. IEEE Ind. Appl. Conf. 41st IAS Annu. Meeting*, vol. 2, Oct. 2006, pp. 988–994, doi: 10.1109/IAS.2006.256645.
- [176] I. Milosavljevic, "Power electronics system communications," M.S. thesis, Virginia Polytech. Inst. State Univ., Blacksburg, VA, USA, 1999.
- [177] Z. Cucej and K. Benkic, "Two optical ring communication between power electronic building blocks: A case study," *Informacije MIDEM-J. Microelectron., Electron. Compon. Mater.*, vol. 38, no. 1, pp. 31–35, 2008.
- [178] Z. Cucej, M. Kaiser, D. Gleich, and P. Planinsic, "Sketch for communication issues in power electronics building blocks connections," in *Proc. IEEE Int. Conf. Ind. Technol.*, vol. 2, Dec. 2003, pp. 1026–1031, doi: 10.1109/ICIT.2003.1290803.
- [179] Ž. Čučej, K. Benkič, M. Milanovič, and M. Truntič, "Inter power electronic building blocks' communication over two optical rings," *IFAC Proc. Volumes*, vol. 41, no. 2, pp. 3328–3332, 2008, doi: 10.3182/20080706-5-KR-1001.00565.
- [180] G. Francis, R. Burgos, F. Wang, and D. Boroyevich, "A power electronics communication protocol for distributed digital control architectures," in *Proc. IEEE Power Energy Soc. Gen. Meeting-Convers. Del. Electr. Energy 21st Century*, Jul. 2008, pp. 1–8, doi: 10.1109/PES.2008.4596827.
- [181] G. Francis, "A synchronous distributed digital control architecture for high power converters," M.S. thesis, Virginia Polytech. Inst. State Univ., Blacksburg, VA, USA, 2004.
- [182] I. Celanovic, "A distributed digital control architecture for power electronics systems," M.S. thesis, Virginia Polytech. Inst. State Univ., Blacksburg, VA, USA, 2000.
- [183] P. C. Loh, D. G. Holmes, and T. A. Lipo, "Implementation and control of distributed PWM cascaded multilevel inverters with minimal harmonic distortion and common-mode voltage," *IEEE Trans. Power Electron.*, vol. 20, no. 1, pp. 90–99, Jan. 2005, doi: 10.1109/TPEL.2004.839830.
- [184] C. L. Toh and L. E. Norum, "A performance analysis of three potential control network for monitoring and control in power electronics converter," in *Proc. IEEE Int. Conf. Ind. Technol.*, Mar. 2012, pp. 224–229, doi: 10.1109/ICIT.2012.6209942.
- [185] J. Wang, H. Wang, and Z.-J. Yang, "An FPGA based slave communication controller for industrial Ethernet," in *Proc. 9th Int. Conf. Solid-State Integr.-Circuit Technol.*, Oct. 2008, pp. 2062–2065, doi: 10.1109/ICSICT.2008.4734979.
- [186] S. Rietmann, S. Fuchs, S. Beck, and J. Biela, "Fieldbus communication scheme for modular converter systems-considerations for minimal switching period and low data latency," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2022, pp. 1–8, doi: 10.1109/ECCE50734.2022.9947450.
- [187] B. Steinmann, G. Fernandez, and N. Cherix, "Tree-shaped networked control system for modular power converters with subµs latency and ns-scale synchronization accuracy," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2019, pp. 6775–6782, doi: 10.1109/ECCE.2019.8912155.

- [188] H. Tu and S. Lukic, "Comparative study of PES net and SyCCo bus: Communication protocols for modular multilevel converter," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2017, pp. 1487–1492, doi: 10.1109/ECCE.2017.8095966.
- [189] Y. Rong, J. Wang, Z. Shen, R. Burgos, D. Boroyevich, and S. Zhou, "Distributed control and communication system for PEBB-based modular power converters," in *Proc. IEEE Electr. Ship Technol. Symp. (ESTS)*, Aug. 2019, pp. 627–633, doi: 10.1109/ESTS.2019. 8847807.
- [190] Y. Rong, J. Wang, Z. Shen, S. Zhou, B. Wen, R. Burgos, D. Boroyevich, J. Verhulst, and M. Belkhayat, "A synchronous distributed communication and control system for SiC-based modular impedance measurement units," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 10, no. 3, pp. 3182–3194, Jun. 2022, doi: 10.1109/JESTPE.2021.3120423.
- [191] V. Mitrovic, B. Fan, Y. Cao, Y. Bai, R. Burgos, and D. Boroyevich, "Distributed communication and control architecture for an intelligent power stage," in *Proc. IEEE 23rd Workshop Control Modeling Power Electron. (COMPEL)*, Jun. 2022, pp. 1–8, doi: 10.1109/COM-PEL53829.2022.9829952.
- [192] Y. Rong, Z. Shen, J. Wang, J. Yu, B. Fan, S. Mocevic, D. Boroyevich, and R. Burgos, "PESNet 3.0: A WRN-based communication network with ±0.5 ns synchronization error for large-scale modular power converters," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 11, no. 2, pp. 1827–1837, Apr. 2023, doi: 10.1109/JESTPE.2022.3219353.
- [193] F. Lotfi, O. Semiari, and F. Afghah, "Evolutionary deep reinforcement learning for dynamic slice management in O-RAN," in *Proc. IEEE Globecom Workshops (GC Wkshps)*, Dec. 2022, pp. 227–232, doi: 10.1109/GCWkshps56602.2022.10008614.
- [194] S. Baba, S. Bachman, M. Jasinski, and H. Li, "Evaluation of modular power converter integrated with 5G network," *Energies*, vol. 14, no. 21, p. 7355, Nov. 2021, doi: 10.3390/en14217355.
- [195] H. Niu and R. D. Lorenz, "Evaluating different implementations of online junction temperature sensing for switching power semiconductors," *IEEE Trans. Ind. Appl.*, vol. 53, no. 1, pp. 391–401, Jan. 2017, doi: 10.1109/TIA.2016.2614773.
- [196] J. Winkler, J. Homoth, and I. Kallfass, "Electroluminescence-based junction temperature measurement approach for SiC power MOSFETs," *IEEE Trans. Power Electron.*, vol. 35, no. 3, pp. 2990–2998, Mar. 2020, doi: 10.1109/TPEL.2019.2929426.
- [197] S. Kalker, C. H. van der Broeck, and R. W. De Doncker, "Utilizing electroluminescence of SiC MOSFETs for unified junctiontemperature and current sensing," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2020, pp. 1098–1105, doi: 10.1109/APEC39645.2020.9124517.
- [198] Y. Avenas, L. Dupont, and Z. Khatir, "Temperature measurement of power semiconductor devices by thermo-sensitive electrical parameters—A review," *IEEE Trans. Power Electron.*, vol. 27, no. 6, pp. 3081–3092, Jun. 2012, doi: 10.1109/TPEL.2011.2178433.
- [199] N. Baker, M. Liserre, L. Dupont, and Y. Avenas, "Junction temperature measurements via thermo-sensitive electrical parameters and their application to condition monitoring and active thermal control of power converters," in *Proc. 39th Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Nov. 2013, pp. 942–948, doi: 10.1109/IECON.2013.6699260.
- [200] C. H. van der Broeck, A. Gospodinov, and R. W. De Doncker, "IGBT junction temperature estimation via gate voltage plateau sensing," *IEEE Trans. Ind. Appl.*, vol. 54, no. 5, pp. 4752–4763, Sep. 2018, doi: 10.1109/TIA.2018.2836362.
- [201] S. Kalker, C. H. van der Broeck, and R. W. De Doncker, "Online junctiontemperature sensing of SiC MOSFETs with minimal calibration effort," in Proc. PCIM Eur. Digit. Days; Int. Exhib. Conf. Power Electron., Intell. Motion, Renew. Energy Energy Manage., Jul. 2020, pp. 1–7.
- [202] C. H. van der Broeck, R. D. Lorenz, and R. W. De Doncker, "Monitoring 3-D temperature distributions and device losses in power electronic modules," *IEEE Trans. Power Electron.*, vol. 34, no. 8, pp. 7983–7995, Aug. 2019, doi: 10.1109/TPEL.2018.2882402.
- [203] C. H. van der Broeck, R. D. Lorenz, and R. W. De Doncker, "Methods for monitoring 3-D temperature distributions in power electronic modules," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2018, pp. 3022–3030, doi: 10.1109/APEC.2018.8341531.
- [204] P. H. Hoang, G. Ozkan, P. R. Badr, L. Timilsina, B. Papari, and C. S. Edrington, "Integrating degradation forecasting into distribution grids' advanced distribution management systems," *Int. J. Electr. Power Energy Syst.*, vol. 150, Aug. 2023, Art. no. 109071, doi: 10.1016/j.ijepes.2023.109071.

- [205] P. H. Hoang, G. Ozkan, P. R. Badr, L. Timilsina, and C. Edrington, "A prognostic based control framework for hybrid electric vehicles," SAE Tech. Paper 2022-01-0352, 2022, doi: 10.4271/2022-01-0352.
- [206] L. Timilsina, P. R. Badr, P. H. Hoang, G. Ozkan, B. Papari, and C. S. Edrington, "Battery degradation in electric and hybrid electric vehicles: A survey study," *IEEE Access*, vol. 11, pp. 42431–42462, 2023, doi: 10.1109/ACCESS.2023.3271287.
- [207] L. Timilsina, P. H. Hoang, A. Arsalan, G. Ozkan, B. Papari, and C. S. Edrington, "Degradation abatement in hybrid electric vehicles using data-driven technique," presented at the 8th Int. Electr. Vehicle Conf. (EVC), Edinburgh, U.K., Jun. 2023.
- [208] L. Timilsina, P. H. Hoang, A. Moghassemi, E. Buraimoh, P. K. Chamarthi, G. Ozkan, B. Papari, and C. S. Edrington, "A real-time prognostic-based control framework for hybrid electric vehicles," *IEEE Access*, 2023.
- [209] A. Arsalan, L. Timilsina, B. Papari, G. Muriithi, G. Ozkan, P. K. Kumar, and C. S. Edrington, "Cyber attack detection and classification for integrated on board electric vehicle chargers subject to stochastic charging coordination," presented at the 8th Int. Electr. Vehicle Conf. (EVC), Edinburgh, U.K., Jun. 2023.
- [210] C. H. Van der Broeck, "Methodology for thermal modeling, monitoring and control of power electronic modules," Ph.D. dissertation, Inst. Power Electron. Elect. Drives, Lehrstuhl und Institut für Stromrichtertechnik und Elektrische Antriebe, RWTH Aachen Univ., Aachen, Germany, 2018, doi: 10.18154/RWTH-2019-01370.
- [211] U.-M. Choi, K. Ma, and F. Blaabjerg, "Validation of lifetime prediction of IGBT modules based on linear damage accumulation by means of superimposed power cycling tests," *IEEE Trans. Ind. Electron.*, vol. 65, no. 4, pp. 3520–3529, Apr. 2018, doi: 10.1109/TIE.2017. 2752142.
- [212] W. Lai, M. Chen, L. Ran, S. Xu, L.-M. Pan, O. Alatise, and P. Mawby, "Study on lifetime prediction considering fatigue accumulative effect for die-attach solder layer in an IGBT module," *IEEJ Trans. Electr. Electron. Eng.*, vol. 13, no. 4, pp. 613–621, Apr. 2018, doi: 10.1002/tee.22607.
- [213] S. M. Avery and R. D. Lorenz, "In situ measurement of wire-bond strain in electrically active power semiconductors," *IEEE Trans. Ind. Appl.*, vol. 49, no. 2, pp. 973–981, Mar. 2013, doi: 10.1109/TIA.2013.2244195.
- [214] M. L. Spencer and R. D. Lorenz, "Analysis and in-situ measurement of thermal-mechanical strain in active silicon power semiconductors," in *Proc. IEEE Ind. Appl. Soc. Annu. Meeting*, Oct. 2008, pp. 1–7, doi: 10.1109/08IAS.2008.360.
- [215] T. A. Polom, B. Wang, and R. D. Lorenz, "Control of junction temperature and its rate of change at thermal boundaries via precise loss manipulation," *IEEE Trans. Ind. Appl.*, vol. 53, no. 5, pp. 4796–4806, Sep. 2017, doi: 10.1109/TIA.2017.2710038.
- [216] D. Kaczorowski, B. Michalak, and A. Mertens, "A novel thermal management algorithm for improved lifetime and overload capabilities of traction converters," in *Proc. 17th Eur. Conf. Power Electron. Appl. (EPE ECCE-Europe)*, Sep. 2015, pp. 1–10, doi: 10.1109/EPE.2015.7309262.
- [217] L. Shao, G. Xu, W. Wei, X. Zhang, H. Li, L. Zheng, and H. Zhao, "Research on junction temperature detection method of power electronic devices based on turn-off time and turn-off loss," *Energy Rep.*, vol. 8, pp. 163–170, Apr. 2022, doi: 10.1016/j.egyr.2021.11.075.
- [218] X. Du, X. Du, J. Zhang, and G. Li, "Numerical junction temperature calculation method for reliability evaluation of power semiconductors in power electronics converters," *J. Power Electron.*, vol. 21, no. 1, pp. 184–194, Jan. 2021, doi: 10.1007/s43236-020-00154-z.
- [219] C. H. van der Broeck and R. W. De Doncker, "Active thermal management for enhancing peak-current capability of three-phase inverters," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2020, pp. 3312–3319, doi: 10.1109/ECCE44975.2020.9235387.
- [220] J. Lemmens, J. Driesen, and P. Vanassche, "Thermal management in traction applications as a constraint optimal control problem," in *Proc. IEEE Vehicle Power Propuls. Conf.*, Oct. 2012, pp. 36–41, doi: 10.1109/VPPC.2012.6422652.
- [221] J. Wölfle, M. Nitzsche, J. Ruthardt, and J. Roth-Stielow, "Junction temperature control system to increase the lifetime of IGBT-powermodules in synchronous motor drives without affecting torque and speed," *IEEE Open J. Power Electron.*, vol. 1, pp. 273–283, 2020, doi: 10.1109/OJPEL.2020.3014443.
- [222] H. Luo, F. Iannuzzo, K. Ma, F. Blaabjerg, W. Li, and X. He, "Active gate driving method for reliability improvement of IGBTs via junction temperature swing reduction," in *Proc. IEEE 7th Int. Symp. Power Electron. Distrib. Gener. Syst. (PEDG)*, Jun. 2016, pp. 1–7, doi: 10.1109/PEDG.2016.7527079.

- [223] C. H. van der Broeck, L. A. Ruppert, R. D. Lorenz, and R. W. De Doncker, "Active thermal cycle reduction of power modules via gate resistance manipulation," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2018, pp. 3074–3082, doi: 10.1109/APEC.2018.8341539.
- [224] H. Lim, J. Hwang, S. Kwon, H. Baek, J. Uhm, and G. Lee, "A study on real time IGBT junction temperature estimation using the NTC and calculation of power losses in the automotive inverter system," *Sensors*, vol. 21, no. 7, p. 2454, Apr. 2021, doi: 10.3390/s21072454.
- [225] D. Kaczorowski and A. Mertens, "Reduction of the EV inverter chip size at constant reliability by active thermal control," in *Proc. IEEE Vehicle Power Propuls. Conf. (VPPC)*, Oct. 2016, p. 1, doi: 10.1109/VPPC.2016.7791759.
- [226] M. Weckert and J. Roth-Stielow, "Chances and limits of a thermal control for a three-phase voltage source inverter in traction applications using permanent magnet synchronous or induction machines," in *Proc. 14th Eur. Conf. Power Electron. Appl.*, Aug. 2011, pp. 1–10.
- [227] M. Weckert and J. Roth-Stielow, "Lifetime oriented control of a threephase voltage source inverter," in *Proc. Power Convers. Intell. Motion* (*PCIM*), 2010, pp. 399–404.
- [228] J. Falck, M. Andresen, and M. Liserre, "Active thermal control of IGBT power electronic converters," in *Proc. 41st Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Nov. 2015, pp. 1–6, doi: 10.1109/IECON.2015.7392925.
- [229] J. Zhang, X. Du, and H.-M. Tai, "A thermal management strategy for smoothing the mission profile thermal cycle of power device in the wind power converter," in *Proc. IEEE Energy Convers. Congr. Expo.* (ECCE), Oct. 2020, pp. 3506–3510, doi: 10.1109/ECCE44975.2020. 9235948.
- [230] P. K. Prasobhu, V. Raveendran, G. Buticchi, and M. Liserre, "Active thermal control of a DC/DC GaN-based converter," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2017, pp. 1146–1152, doi: 10.1109/APEC.2017.7930840.
- [231] H. Song, I. Cvetkovic, R. Zhang, C. DiMarino, and D. Boroyevich, "Gate driver switching noise propagation study for medium voltage SiC-based power electronics building blocks," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2022, pp. 1–8, doi: 10.1109/ECCE50734.2022.9948024.
- [232] A. Soldati, F. Dossena, G. Pietrini, D. Barater, C. Concari, and F. Iannuzzo, "Thermal stress mitigation by active thermal control: Architectures, models and specific hardware," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2017, pp. 3822–3829, doi: 10.1109/ECCE.2017.8096674.
- [233] A. Soldati, C. Concari, F. Dossena, D. Barater, F. Iannuzzo, and F. Blaabjerg, "Active thermal control for reliability improvement of MOS-gated power devices," in *Proc. 43rd Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Oct. 2017, pp. 7935–7940, doi: 10.1109/IECON.2017.8217391.
- [234] D. Qin, G. Ozkan, C. Edrington, and Z. Zhang, "Electrothermal management using in-situ junction temperature monitoring for enhanced reliability of SiC-based power electronics," in *Proc. IEEE Electr. Ship Technol. Symp. (ESTS)*, Aug. 2021, pp. 1–7, doi: 10.1109/ESTS49166.2021.9512370.
- [235] S. Mocevic, V. Mitrovic, J. Wang, R. Burgos, and D. Boroyevich, "Gate-driver integrated junction temperature estimation of SiC MOSFET modules," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 10, no. 5, pp. 4965–4980, Oct. 2022, doi: 10.1109/JESTPE.2021.3108442.
- [236] M. Saur, B. Piepenbreier, W. Xu, and R. D. Lorenz, "Implementation and evaluation of inverter loss modeling as part of DB-DTFC for loss minimization each switching period," in *Proc. 16th Eur. Conf. Power Electron. Appl.*, Aug. 2014, pp. 1–10, doi: 10.1109/EPE.2014. 6910691.
- [237] M. Andresen, G. Buticchi, and M. Liserre, "Thermal stress analysis and MPPT optimization of photovoltaic systems," *IEEE Trans. Ind. Electron.*, vol. 63, no. 8, pp. 4889–4898, Aug. 2016, doi: 10.1109/TIE.2016.2549503.
- [238] J. Wölfle, O. Lehmann, and J. Roth-Stielow, "A novel control method to improve the reliability of traction inverters for permanent magnet synchronous machines," in *Proc. IEEE 11th Int. Conf. Power Electron. Drive Syst.*, Jun. 2015, pp. 379–384, doi: 10.1109/PEDS.2015. 7203573.
- [239] X. Wang, A. Castellazzi, and P. Zanchetta, "Observer based temperature control for reduced thermal cycling in power electronic cooling," *Appl. Thermal Eng.*, vol. 64, nos. 1–2, pp. 10–18, Mar. 2014, doi: 10.1016/j.applthermaleng.2013.12.001.

- [240] Y. Yerasimou, V. Pickert, B. Ji, and X. Song, "Liquid metal magnetohydrodynamic pump for junction temperature control of power modules," *IEEE Trans. Power Electron.*, vol. 33, no. 12, pp. 10583–10593, Dec. 2018, doi: 10.1109/TPEL.2018.2806622.
- [241] C. Li, D. Jiao, J. Jia, F. Guo, and J. Wang, "Thermoelectric cooling for power electronics circuits: Modeling and active temperature control," *IEEE Trans. Ind. Appl.*, vol. 50, no. 6, pp. 3995–4005, Nov. 2014, doi: 10.1109/TIA.2014.2319576.
- [242] Y. Ko, M. Andresen, G. Buticchi, and M. Liserre, "Discontinuousmodulation-based active thermal control of power electronic modules in wind farms," *IEEE Trans. Power Electron.*, vol. 34, no. 1, pp. 301–310, Jan. 2019, doi: 10.1109/TPEL.2018.2819423.
- [243] J. Wolfle, M. Nitzsche, J. Weimer, M. Stempfle, and J. Roth-Stielow, "Temperature control system using a hybrid discontinuous modulation technique to improve the lifetime of IGBT power modules," in *Proc. 18th Eur. Conf. Power Electron. Appl. (EPE ECCE Europe)*, Sep. 2016, pp. 1–10, doi: 10.1109/EPE.2016.7695403.
- [244] J. Falck, M. Andresen, and M. Liserre, "Thermal-based finite control set model predictive control for IGBT power electronic converters," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2016, pp. 1–7, doi: 10.1109/ECCE.2016.7855513.
- [245] G. Ozkan, P. Hoang, P. Badr, L. Timilsina, C. Edrington, Q. Zhu, Y. Parvini, and R. Prucka, "Model-based active thermal management for neutral-point clamped power converter with adaptive weight," SAE Tech. Paper 2022-01-0727, 2022.
- [246] G. Ozkan, P. H. Hoang, P. R. Badr, C. S. Edrington, and B. Papari, "Real-time thermal management for two-level active rectifier with finite control set model predictive control," *Int. J. Electr. Power Energy Syst.*, vol. 131, Oct. 2021, Art. no. 107057, doi: 10.1016/j.ijepes.2021. 107057.
- [247] G. Ozkan, P. Hoang, P. Badr, and C. Edrington, "Electro-thermal control on power electronic converters: A finite control set model predictive control approach," *SAE Int. J. Adv. Current Practices Mobility*, vol. 4, no. 1, pp. 151–156, Apr. 2021, doi: 10.4271/2021-01-0200.
- [248] M. Novak, V. N. Ferreira, M. Andresen, T. Dragicevic, F. Blaabjerg, and M. Liserre, "FS-MPC algorithm for optimized operation of a hybrid active neutral point clamped converter," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2019, pp. 1447–1453, doi: 10.1109/ECCE.2019.8913306.
- [249] J. Brandelero, J. Ewanchuk, N. Degrenne, and S. Mollov, "Lifetime extension through Tj equalisation by use of intelligent gate driver with multi-chip power module," *Microelectron. Rel.*, vols. 88–90, pp. 428–432, Sep. 2018, doi: 10.1016/j.microrel.2018.07.034.
- [250] V. Ferreira, M. Andresen, B. Cardoso, and M. Liserre, "Selective softswitching for thermal balancing in IGBT-based multichip systems," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 9, no. 4, pp. 3982–3991, Aug. 2021, doi: 10.1109/JESTPE.2020.3026782.
- [251] J. Ewanchuk, J. Brandelero, and S. Mollov, "Lifetime extension of a multi-die SiC power module using selective gate driving with temperature feedforward compensation," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2017, pp. 2520–2526, doi: 10.1109/ECCE.2017.8096480.
- [252] J. Ewanchuk, J. Brandelero, and S. Mollov, "A gate driver based approach to improving the current density in a power module by equalizing the individual die temperatures," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2018, pp. 4652–4658, doi: 10.1109/ECCE.2018. 8557880.
- [253] M. Sasaki, H. Nishio, A. Shorten, and W. T. Ng, "Current balancing control for parallel connected IGBTs using programmable gate driver output resistance," in *Proc. 25th Int. Symp. Power Semiconductor Devices IC's (ISPSD)*, May 2013, pp. 65–68, doi: 10.1109/ISPSD.2013. 6694398.
- [254] Y. Dong, H. Yang, W. Li, and X. He, "Neutral-point-shift-based active thermal control for a modular multilevel converter under a singlephase-to-ground fault," *IEEE Trans. Ind. Electron.*, vol. 66, no. 3, pp. 2474–2484, Mar. 2019, doi: 10.1109/TIE.2018.2833019.
- [255] R. Han, Q. Xu, H. Ding, P. Guo, J. Hu, Y. Chen, and A. Luo, "Thermal stress balancing oriented model predictive control of modular multilevel switching power amplifier," *IEEE Trans. Ind. Electron.*, vol. 67, no. 11, pp. 9028–9038, Nov. 2020, doi: 10.1109/TIE.2019.2956384.
- [256] M. Aly, G. M. Dousoky, and M. Shoyama, "Lifetime-oriented SVPWM for thermally-overloaded power devices in three-level inverters," in *Proc. 41st Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Nov. 2015, pp. 3614–3619, doi: 10.1109/IECON.2015.7392662.

- [257] S. Mocevic, J. Yu, B. Fan, J. Stewart, Y. Rong, V. Mitrovic, R. Burgos, and D. Boroyevich, "High power density 10 kV SiC MOSFET-based modular, scalable converter for medium voltage applications," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2022, pp. 422–429, doi: 10.1109/APEC43599.2022.9773720.
- [258] M. Liserre, V. Raveendran, and M. Andresen, "Graph-theory-based modeling and control for system-level optimization of smart transformers," *IEEE Trans. Ind. Electron.*, vol. 67, no. 10, pp. 8910–8920, Oct. 2020, doi: 10.1109/TIE.2019.2947837.
- [259] H. Wang, A. M. Khambadkone, and X. Yu, "Control of parallel connected power converters for low voltage microgrid—Part II: Dynamic electrothermal modeling," *IEEE Trans. Power Electron.*, vol. 25, no. 12, pp. 2971–2980, Dec. 2010, doi: 10.1109/TPEL.2010.2087394.
- [260] V. Raveendran, M. Andresen, M. Liserre, and G. Buticchi, "Lifetimebased power routing of smart transformer with CHB and DAB converters," in *Proc. IEEE Appl. Power Electron. Conf. Expo. (APEC)*, Mar. 2018, pp. 3523–3529, doi: 10.1109/APEC.2018.8341612.
- [261] V. Raveendran, M. Andresen, and M. Liserre, "Lifetime control of modular smart transformers considering the maintenance schedule," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2018, pp. 60–66, doi: 10.1109/ECCE.2018.8557905.
- [262] V. N. Ferreira, R. R. Bastos, T. S. de Souza, M. Liserre, and B. J. C. Filho, "Power routing to enhance the lifetime of multiphase drives," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2019, pp. 3215–3222, doi: 10.1109/ECCE.2019.8912972.
- [263] T. Ericsen, Y. Khersonsky, and P. K. Steimer, "PEBB concept applications in high power electronics converters," in *Proc. IEEE* 36th Conf. Power Electron. Spec., Jun. 2005, pp. 2284–2289, doi: 10.1109/PESC.2005.1581950.
- [264] P. K. Steimer, O. Apeldoorn, B. Odegard, S. Bernet, and T. Bruckner, "Very high power PEBB technology," in *Proc. Eur. Conf. Power Electron. Appl.*, 2005, p. 11, doi: 10.1109/EPE.2005.219729.
- [265] A. M. Alcaide, G. Buticchi, A. Chub, and L. Dalessandro, "Design and control for high reliability power electronics: State of the art and future trends," *IEEE J. Emerg. Sel. Topics Ind. Electron.*, early access, Jun. 20, 2023, doi: 10.1109/JESTIE.2023.3287513.
- [266] A. Bakeer, A. Chub, and D. Vinnikov, "Full-bridge fault-tolerant isolated DC–DC converters: Overview of technologies and application challenges," *IEEE Power Electron. Mag.*, vol. 9, no. 3, pp. 45–55, Sep. 2022, doi: 10.1109/MPEL.2022.3196565.
- [267] H. A. Mantooth, M. D. Glover, and P. Shepherd, "Wide bandgap technologies and their implications on miniaturizing power electronic systems," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 2, no. 3, pp. 374–385, Sep. 2014, doi: 10.1109/JESTPE.2014.2313511.
- [268] J. O. Gonzalez and O. Alatise, "Bias temperature instability and junction temperature measurement using electrical parameters in SiC power MOS-FETs," *IEEE Trans. Ind. Appl.*, vol. 57, no. 2, pp. 1664–1676, Mar. 2021, doi: 10.1109/TIA.2020.3045120.
- [269] A. Stippich, C. H. Van Der Broeck, A. Sewergin, A. H. Wienhausen, M. Neubert, P. Schülting, S. Taraborrelli, H. van Hoek, and R. W. De Doncker, "Key components of modular propulsion systems for next generation electric vehicles," *CPSS Trans. Power Electron. Appl.*, vol. 2, no. 4, pp. 249–258, Dec. 2017, doi: 10.24295/CPSSTPEA.2017.00023.
- [270] A. Stippich, C. H. van der Broek, and R. W. De Doncker, "Enhancing lifetime of power electronic modules via thermal buffers," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2020, pp. 3178–3185, doi: 10.1109/ECCE44975.2020.9236041.
- [271] P. K. Prasobhu, G. Buticchi, S. Brueske, and M. Liserre, "Gate driver for the active thermal control of a DC/DC GaN-based converter," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Sep. 2016, pp. 1–8, doi: 10.1109/ECCE.2016.7855131.
- [272] Z. Ni, S. Zheng, M. S. Chinthavali, and D. Cao, "Investigation of dynamic temperature-sensitive electrical parameters for medium-voltage SiC and Si devices," *IEEE J. Emerg. Sel. Topics Power Electron.*, vol. 9, no. 5, pp. 6408–6423, Oct. 2021, doi: 10.1109/JESTPE.2021.3054018.
- [273] Z. Ni, S. Zheng, M. S. Chinthavali, and D. Cao, "Investigation of dynamic temperature-sensitive electrical parameters for mediumvoltage low-current silicon carbide and silicon devices," in *Proc. IEEE Energy Convers. Congr. Expo. (ECCE)*, Oct. 2020, pp. 3376–3382, doi: 10.1109/ECCE44975.2020.9236121.
- [274] A. Griffo, J. Wang, K. Colombage, and T. Kamel, "Real-time measurement of temperature sensitive electrical parameters in SiC power MOSFETs," *IEEE Trans. Ind. Electron.*, vol. 65, no. 3, pp. 2663–2671, Mar. 2018, doi: 10.1109/TIE.2017.2739687.

- [275] D. L. Blackburn, "Temperature measurements of semiconductor devices—A review," in *Proc. 20th Annu. IEEE Semiconductor Thermal Meas. Manage. Symp.*, Mar. 2004, pp. 70–80, doi: 10.1109/STHERM.2004.1291304.
- [276] T. Thomas, K.-F. Becker, T. Braun, M. van Dijk, O. Wittler, and K.-D. Lang, "Assessment of high temperature reliability of molded smart power modules," in *Proc. 5th Electron. Syst.-Integr. Technol. Conf.* (*ESTC*), Sep. 2014, pp. 1–7, doi: 10.1109/ESTC.2014.6962863.
- [277] J. Schuderer, V. Lindstroem, C. Liu, and F. Mohn, "Transfer molding for power semiconductor modules," in *Proc. 10th Int. Conf. Integr. Power Electron. Syst. (CIPS)*, Mar. 2018, pp. 1–6.
- [278] C. Marczok, M. Martina, M. Laumen, S. Richter, A. Birkhold, B. Flieger, O. Wendt, and T. Paesler, "SiCmodul-modular high-temperature SiC power electronics for fail-safe power control in electrical drive engineering," in *Proc. 11th Int. Conf. Integr. Power Electron. Syst. (CIPS)*, Mar. 2020, pp. 1–6.



**ALI MOGHASSEMI** (Graduate Student Member, IEEE) received the B.S. and M.S. degrees in electrical engineering from Islamic Azad University, South Tehran Branch, Tehran, Iran, in 2012 and 2015, respectively. He is currently pursuing the Ph.D. degree with the Real-Time COntrol and Optimization Laboratory (RT-COOL), Holcombe Department of Electrical and Computer Engineering, Clemson University, Clemson, SC, USA, under the supervision of Dr. Christopher S.

Edrington and Dr. Gokhan Ozkan. He is a Graduate Research Assistant of electrical engineering with the Real-Time COntrol and Optimization Laboratory (RT-COOL), Holcombe Department of Electrical and Computer Engineering, Clemson University. He has authored and coauthored several peer-reviewed papers in high-quality journals and conferences, two book chapters, and one book. His research interests include modeling, simulation, and real-time modeling of power electronics and renewable energy systems. He is currently working on electro-thermal modeling, simulation, and control of power electronics building blocks (PEBBs) in all-electric ships (AESs). He is serving as a volunteer reviewer for several scientific journals/conferences over the years.



**S. M. IMRAT RAHMAN** (Graduate Student Member, IEEE) received the B.Sc. degree in electrical and electronic engineering from the Islamic University of Technology, Bangladesh, in 2010, and the M.Sc. degree in power engineering from the Technical University of Munich, Germany, in 2015. He is currently pursuing the Ph.D. degree with the Holcombe Department of Electrical and Computer Engineering (ECE), Clemson University, under the supervision of

Dr. Christopher S. Edrington and Dr. Gokhan Ozkan. He is a Graduate Research Assistant with the Real-Time COntrol and Optimization Laboratory (RT-COOL). His research interests include the active thermal management of power electronic converters, distributed control, and thermal modeling of power electronic building blocks (PEBB) systems.



**GOKHAN OZKAN** (Senior Member, IEEE) received the B.S. degree in teacher training and in electrical field and energy system engineering from Marmara University and Erciyes University, Turkey, in 2006 and 2014, respectively, the M.S. degree in energy system engineering from Erciyes University, in 2015, and the Ph.D. degree in electrical and computer engineering from the FAMU-FSU College of Engineering. He was a Lecturer with Bozok University, Turkey. He was a Graduate

Research Assistant with the Center for Advanced Power Systems. He joined as a Postdoctoral Research Associate with the RT-COOL Group, Clemson University, where he is currently a Research Assistant Professor. His research interests include renewable energy sources, power and energy management, real-time modeling and simulation, power electronics and control, distributed energy resources, and electrical vehicles.



**CHRISTOPHER S. EDRINGTON** (Senior Member, IEEE) received the B.S. degree in engineering from Arkansas State University, in 1999, and the M.S. and Ph.D. degrees in electrical engineering from the Missouri University of Science and Technology, in 2001 and 2004, respectively. He was a DoE GAANN, NSF IGERT, and Grainger Foundation Fellow with the Missouri University of Science and Technology. He is currently a Warren H. Owen Distinguished Professor of electrical and

computer engineering with Clemson University. He is the Lead of the Real-Time COntrol and Optimization Laboratory (RT-COOL) and the Co-Director of the Smart Energy and Propulsion Focus Area, VIrtual PRototyping for Ground Vehicle Systems (VIPR-GS) Center. He has published more than 170 papers (including two IEEE Prize Awards and multiple conference paper awards), has graduated 23 M.S. students and 16 Ph.D. students (with three in the process), and has eight patents (in real-time stability and complexity metrics, insulation breakdown, and linear machines). His research interests include modeling, simulation, and control of electromechanical drive systems; applied power electronics; distributed control; integration of renewable energy, storage, and pulse power loads.



**ZHEYU ZHANG** (Senior Member, IEEE) is currently a Warren H. Owen–Duke Energy Assistant Professor with the Holcombe Department of Electrical and Computer Engineering, Clemson University, Clemson, SC, USA. He has published more than 100 papers, filed more than ten patent applications, and presented ten IEEE tutorial seminars and webinars. His research interests include silicon carbide-based power electronics characterization and applications. He received the 2021

IEEE Industry Application Society (IAS) Andrew W. Smith Outstanding Young Member Achievement Award.



**PHANI KUMAR CHAMARTHI** (Member, IEEE) received the joint M.Tech. and Ph.D. degrees with a specialization in power electronics and power systems from the Department of Electrical and Electronics Engineering, Indian Institute of Technology Bombay, Mumbai, India, in 2018. He is currently a Postdoctoral Fellow with Clemson University, USA. His research interests include new single-stage transformerless inverter topologies, multilevel converter topologies, space vector

control of multilevel converters, and new modulation and control strategies of multilevel converters, focusing on renewable energy applications.

....